

Initiated by Deutsche Post Foundation

# DISCUSSION PAPER SERIES

IZA DP No. 11234

Long-Term Effects of Childhood Nutrition: Evidence from a School Lunch Reform

Jesper Alex-Petersen Petter Lundborg Dan-Olof Rooth

DECEMBER 2017



Initiated by Deutsche Post Foundation

# DISCUSSION PAPER SERIES

IZA DP No. 11234

Long-Term Effects of Childhood Nutrition: Evidence from a School Lunch Reform

#### Jesper Alex-Petersen Lund University

**Petter Lundborg** Lund University and IZA

Dan-Olof Rooth SOFI, Stockholm University and IZA

DECEMBER 2017

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

| IZA – Institute of Labor Economics |                             |             |
|------------------------------------|-----------------------------|-------------|
| Schaumburg-Lippe-Straße 5–9        | Phone: +49-228-3894-0       |             |
| 53113 Bonn, Germany                | Email: publications@iza.org | www.iza.org |

# ABSTRACT

# Long-Term Effects of Childhood Nutrition: Evidence from a School Lunch Reform\*

We examine the long-term impact of a policy that introduced free and nutritious school lunches in Swedish primary schools. For this purpose, we use historical data on the gradual implementation of the policy across municipalities and employ a difference-in-differences design to estimate the impact of this lunch policy on a broad range of medium and long-term outcomes, including lifetime income, health, cognitive skills, and education. Our results show that the school lunch program generated substantial long-term benefits, where pupils exposed to the program during their entire primary school period have 3 percent greater life-time earnings. In addition, we find the effect to be greater for pupils that were exposed at earlier ages and for pupils from poor households. Finally, exposure to the school lunch program had substantial effects on educational attainment and health and these effects can explain a large part of the return to school lunches.

| JEL Classification: | I12, I38, J24   |
|---------------------|---|
| Keywords:           | nutrition, early life, childhood, long-term, income, causal |

### **Corresponding author:**

Petter Lundborg Department of Economics Lund University SE-220 07 Lund Sweden E-mail: petter.lundborg@nek.lu.se

<sup>\*</sup> We are thankful for comments by Douglas Almond, Hillary Hoynes, Gordon Dahl, Michael Kremer, Bhaskar Mazumder, Erik Plug, Kirabo Jackson, Peter Nilsson, Robert Östling, Pia Pinger, Randi Hjalmarsson, and seminar participants at IIES, the Frisch Centre at University of Oslo, Academia Sinica, Aix-Marseille School of Economics, Lund University, Gothenburg University, SOFI at Stockholm University, and the EALE, ESPE, and RES conferences. All opinions and any errors are our own. The data used in this paper come from the Swedish Interdisciplinary Panel (SIP) administered at the Centre for Economic Demography, Lund University, Sweden.

# 1 Introduction

Recent research has shown beneficial long-term effects of early childhood interventions (Currie & Almond 2011a; Currie & Almond 2011b). We know much less, however, about the long-term effects of government policies targeting children in the phase between early childhood and adulthood, such as school meal programs (Almond et al. 2017). The period between childhood and adolescence is believed to be a critical period for diets of high nutritional quality and, today, many Western countries feed their school-age children through extensive government sponsored school-meal programs (WHO 2006). In the U.S., for instance, the National School Lunch Program (NSLP) alone serves approximately 30 million children (57% of the student body) at a cost of \$13.6 billion.<sup>1</sup>

Despite school lunch programs being around since the 1940s in rich countries such as Sweden and the U.S., it has been difficult to evaluate these programs' effectiveness. Since the U.S. school lunch program is federal, with no or little variation across areas, commonly used quasi-experimental approaches are not easily applied (Hoynes & Schanzenbach 2015). Hence, it is unclear how effective school lunch programs are relative to other related social reforms, such as food stamps, HEAD start, or increasing income transfers to poor families, in improving long-term outcomes (Ludwig & Miller 2007; Dahl & Lochner 2012; Hoynes et al. 2016). Moreover, the quality of school meal programs has recently been questioned in many Western countries, where the increasing prevalence of overweight and obese children has put school lunches at top of the debate (Schanzenbach 2009). This has led governments to impose stricter nutritional standards on school lunches but evidence on the long-term impacts of such initiatives is lacking.

This study provides evidence on the medium and long-term economic benefits of an extensive school lunch program. More specifically, we ask whether a policy that introduced free and nutritious school lunches in Swedish primary schools in the 1950s and 1960s improved children's medium and long-term economic, cognitive, educational, and health outcomes. The policy imposed strict nutritional standards on the meals served, which were to provide a third of the daily caloric need; contain strictly specified amounts of proteins, vitamins, calcium and iron; and contain a maximum fat content. Interestingly, these standards are very similar to the ones introduced in more recent meal programs, such as the "School Meals Initiative for Healthy Children" that was passed by the U.S. congress in 1995 in response to evidence questioning the nutritional quality of the NSLP program.<sup>2</sup> Evaluating the Swedish program therefore constitutes a unique opportunity to improve our understanding of the potential long-term effects of more recent initiatives to serve children more healthy school meals.<sup>3</sup>

The main motivation behind the Swedish school lunch program was that the nutritional

 $<sup>\</sup>label{eq:linear} \ensuremath{^1https://www.ers.usda.gov/topics/food-nutrition-assistance/child-nutrition-programs/national-school-lunch-program/.$ 

<sup>&</sup>lt;sup>2</sup>The nutritional standards of the NLSP were further updated in 2012 under the "Hunger-Free Kids Act" but the Trump administration recently scaled back these standards in a proclamation signed by the Secretary of Agriculture Sonny Perdue (see http://edition.cnn.com/2017/05/02/health/school-lunch-changes/index.html).

<sup>&</sup>lt;sup>3</sup>The Swedish school lunch program also shares other characteristics with recent programs, including (partial) reimbursement of costs to schools, controls that schools follow the program, and the provision of education and training of school food personnel.

quality of the meals consumed during the school lunch break was deemed inadequate and, in particular, fell short in terms of vitamins A, C and D, protein, and certain minerals. Food shortage and hunger was uncommon in Sweden during the 1950s and 1960s and the program was aimed at improving nutritional standards rather than improving access to food or caloric intake. Another motivation behind the policy was to ease the economic burden on the households by freeing them from the task of providing school lunches (SOU 1945). An interesting feature of the program was therefore that it provided improved nutrition, but potentially also improved household finances.

To estimate the impact of the program, we use newly collected historical data on its gradual implementation across municipalities in Sweden between the years 1959 and 1969. During this time period, almost 300 municipalities introduced the program, with a fairly equal number of municipalities per year. We have linked the data on the program's introduction to administrative records on approximately 1.5 million individuals, covering the population of primary school pupils during the study period. Using a difference-in-differences design, we then estimate the impact of the program on a broad range of outcomes measured from a variety of registers, including income and education registers, the military enlistment register, the medical birth register, and hospitalization and mortality registers.

Our results show that the Swedish school lunch program generated substantial long-term benefits, where pupils exposed to the program during their entire primary school period have 3 percent greater life-time earnings compared to pupils who were never exposed. We also find interesting heterogeneity in the effects, where children from poor households benefit the most from the program, although children from all households, except the richest, benefit to some extent. In addition, we show evidence of a clear dose-response relationship, where earlier exposure is associated with greater effects. Importantly, the school lunch program did not affect the probability of being obese or overweight at age 18, suggesting that the school lunch program did not provide an unhealthy excess of calories.

To shed light on possible mechanisms, we use data from multiple sources. Using data from the education register, we find that greater exposure to school meals had large and positive effects on years of schooling that can explain one half of the income effect. Using data from the military enlistment records and the medical birth register, we find that males and females who were more exposed to the meals were taller and that males were assessed to be healthier at time of enlistment, suggesting that improved nutrition, which was specifically at focus in the school lunch program, is an important mechanism. Using data from the hospitalization and mortality registers, we find no long-term effects on mortality or morbidity, however.

Since the school lunches were provided free of charge, an alternative mechanism behind our results is improvements in household income due to reduced food expenditures. Moreover, the school meal program may have affected female labor supply since it was no longer necessary for mothers to stay home and cook during the lunch break. When we run our analysis on closely spaced siblings, who should all be exposed to such potential improvements in household income, the estimates remain at least as strong. Moreover, when we calculate the potential savings in food expenditures of the household from free school lunches, we conclude that they are too small to generate long-term effects. School lunches may also affect earnings through school attendance. In the Swedish context, we can rule out such effects since primary school was mandatory and since attendance rates were high already before the lunch program was introduced.

We address a number of threats to our empirical design. To assess the parallel trend assumption, we compare pretreatment trends across early, mid, and late adopters of the reform and find that they are parallel. In addition, an event-study analysis show that the effect arises for cohorts exposed to the lunch program but not for unexposed cohorts. The results are also robust to the inclusion of linear and quadratic municipality-specific trends. In addition, we check for any systematic migration of families in response to the program and find no evidence for such strategic behavior. The main results are also insensitive to using alternative income measures and to controlling for other potentially important reforms that were rolled out during the same time period.

Our paper makes several contributions to the literature. First, it fills a gap in the literature on the effect of early life policies on long-term outcomes. While previous studies have established that policies such as food stamps can have important long-term effects for those exposed in utero, we know less about the effect of policy-driven improvements during later phases of childhood and adolescence. Knowing if there are critical periods during which interventions are particularly effective is useful for policy-makers (van den Berg et al. 2014; Chetty et al. 2016; Hoynes et al. 2016).

Second, we provide new and rare evidence on the long-term effects of a policy-driven improvement in early-life nutrition. Most of the existing evaluations of school meal programs are of a short-term nature and evaluate meal programs where the food served is of questionable nutritional quality.<sup>4</sup> Third, by focusing on a policy that affected all children in school, we can investigate heterogeneity in the effect of the school lunch program across the entire parental income distribution. Fourth, whereas many of the policies studied previously, such as food stamps and welfare benefits, affect only a subset of the population, and therefore can be stigmatizing and create incentives for parents to modify their behavior in order to remain eligible for the program benefit, we study a universal policy where such effects can be ruled out.

The paper proceeds as follows: In the next section, we discuss the literature on the longterm effects of early life policies, including school meal policies, and discuss the different channels through which school meal programs may affect long-term outcomes. Section 3 describes the Swedish policy and setting in more detail. Section 4 describes our data, and Section 5 introduces our empirical design. Section 6 shows our main results whereas Section 7 shows the results on potential mechanisms through which the long-term effects may appear. Section 8 show the results across generations and Section 9 provides a set of robustness checks. Section 10 provides a cost-benefit calculation of the school-lunch program, and Section 11 concludes.

<sup>&</sup>lt;sup>4</sup>We review the relevant literature in Section II.

# 2 Literature background

#### 2.1 Early life policies

The negative long-term effects of various shocks in utero or infancy are now well established, and the economic literature has instead turned to analyzing the long-term effects of early life *policies*. Since this latter literature is closely related to our paper, we briefly review some of it below before turning to studies that have specifically analyzed the effect of school meal policies.<sup>5</sup>

An important literature evaluates the long-term effects of various welfare programs specifically targeted toward poor households. Hoynes et al. (2016) analyze the introduction of the Food Stamp Program across counties in the U.S. and show that the program led to a reduction in the incidence of metabolic syndrome and to increased women's self-sufficiency for cohorts that were exposed in childhood (ages 0-5). The authors argue that the program should mainly be viewed as an income transfer program rather than as a nutrition program. Earlier work by the same authors showed health gains at birth for babies whose mothers had access to food stamps during pregnancy, thus pointing to nutrition as an important pathway for the effects; see Hoynes et al. (2011).<sup>6</sup> Aizer et al. (2016) show that children in poor families who had access to cash transfers had favorable outcomes in terms of mortality, education, and earnings, and Ludwig & Miller (2007) show that the introduction of the Head Start program to poor children aged three to five (and their parents) increased educational attainment and reduced child mortality rates. Dahl & Lochner (2012) study the effect of the earned income tax credit on child outcomes and find positive effects on children's test scores, and Milligan & Stabile (2011) find similar effects in Canada.

In addition, more specific policies with the potential to affect fetuses, infants and small children have proven to be important for later life outcome. Nilsson (2017) shows that increased access to alcohol during pregnancy has negative long-term effects on affected children's earnings and education during adulthood. Other papers find positive long-term effects of specific health interventions such as treatments for low birth weight (Almond et al. 2010; Bharadwaj et al. 2013), breastfeeding interventions (Fitzsimons & Vera-Hernández 2016), improved access to maternal health clinics (Bhalotra et al. 2017; Hjort et al. 2017), expansions of Medicaid (Wherry & Meyer 2016; Brown et al. 2015), improved access to hospitals (Chay et al. 2009), and improved access to water fluoridation (Neidell et al. 2010).<sup>7</sup>

Most of the papers mentioned above concern policies affecting children in utero or the first years of life. For later phases of childhood, there is still very limited evidence and our paper extends the literature by focusing on the long-term effects of a policy targeting children at ages 7 to 16. A notable exception is Chetty et al. (2016), who analyze the long-term impact of the Moving-To-Opportunity (MTO) Program and show that children who were exposed to the program, and therefore moved to a lower-poverty area, before the age

<sup>&</sup>lt;sup>5</sup>For a thorough literature review on the effect of early life policies in developed countries, see Almond et al. (2017).

<sup>&</sup>lt;sup>6</sup>Nutritional interventions have been studied in the context of developing countries. Hoddinott et al. (2008) and Maluccio et al. (2009) studied the effect of a childhood nutritional intervention in Guatemala and found large and positive long-term effects on educational outcomes and adult income.

 $<sup>^{7}</sup>$ Other papers evaluate policies that indirectly affect early life health, such as measure to reduce pollution (Nilsson 2009; Isen et al. 2017).

of 13, faced substantial long-term economic benefits.<sup>89</sup>

#### 2.2 School lunch policies

Most of the literature on the impact of school meals focuses on various programs in the United States, such as the National School Lunch Program (NSLP) or various school breakfast programs.<sup>10</sup> It should be noted that the Swedish school lunch program differed from the historical NSLP in several ways. First, while the Swedish program provided specific nutrition guidance, as described in Section 3 below, similar U.S. guidelines were introduced as late as 1995. In line with the lack of specific nutrition guidelines, several early analyses of the nutritional content of the NSLP lunches revealed that NSLP participants were not nutritionally better off compared to non-participants (Paige 1972; Hanes et al. 1984). Later evaluations of the NSLP found that the program increased the intake of fat and important minerals, vitamins, and proteins but no long-term evaluations of this later phase of NSLP are available (see Hopkins & Gunther (2015) for a literature review). Second, the Swedish school lunches were, and still are, served free of charge for all pupils whereas the NSLP was, and still is, subsidized only for low-income families.

The findings regarding the effect of the NSLP are mixed and almost exclusively of a short-term nature. In a recent literature overview on the effect of various U.S. food and nutrition programs, Hoynes & Schanzenbach (2015) noted: "We have much more to learn about the potential benefits of these programs on health and wellbeing in the long run, and when in the life cycle is the most important time to provide these benefits."

Several studies have analyzed whether the NSLP contributes to the overweight and obesity "epidemic". Schanzenbach (2009) exploits sharp discontinuities in eligibility for reduced-price lunch to show that childhood obesity increases with exposure to the NSLP program. Similar findings were reported by Millimet et al. (2010), using panel data, but studies by Mirtcheva & Powell (2013) and Dunifon & Kowaleski-Jones (2004), also using panel data, and Gundersen et al. (2012), using a non-parametric bounds analysis, find opposite or no effects.

Short-term educational outcomes were analyzed in a U.K. study by Belot & James (2011), who evaluated the "Feed Me Better" campaign run by the British celebrity chef Jamie Oliver. The program aimed at improving the nutritional standards of school lunches in the U.K and the results showed a short-term decrease in absences and improved school grades following the introduction of the program. Using data from Californian public schools, Anderson et al. (2017) find positive effects on student achievement tests for students at schools who had contracts with healthy school vendors. Analyzing the short-term effects of a school lunch program in Chile aimed at increasing caloric intake, McEwan (2013) found no effects on

<sup>&</sup>lt;sup>8</sup>Oreopoulos (2003) found weaker effects when studying differences in long-term outcomes between children assigned to substantially different housing projects in Toronto. The mean age at assignment was higher than in the MTO experiment, however. Jacob et al. (2015) found no long-term effects on child outcomes from a program that randomized housing vouchers in Chicago.

 $<sup>^{9}</sup>$ In the economics of education literature a number of papers have analyzed the long-term impact of various educational interventions such as class size reductions; see Burgess (2016).

 $<sup>^{10}</sup>$  See Galer-Unti et al. (1995) for a discussion on the introduction of NSLP.

attendance or grades.

We are aware of only one paper that evaluates the long-term impact of free school lunches. Hinrichs (2010) uses a change in the formula used by the federal government to allocate funding to the states in order to study the long-term impact of exposure to the NSLP program. He finds no effects on health but finds large and positive effects on educational attainment, where an increase in NSLP exposure in a state by 10 percentage points increases completed education by nearly one year on average for males and one-third of a year for females. As discussed by Hinrichs (2010), the impact of the NSLP on educational attainment may reflect increased attendance in schools following the access to school lunches together with an increase in calories consumed. To the best of our knowledge, there are no studies that estimate the long-term impact of a policy specifically targeted at increasing the quality of the school lunch.

Another related strand of literature evaluates the impact of various school *breakfast* programs, again mostly in the short run. Several studies report short-term positive effects of the program in terms of dietary quality, reduced obesity, school achievement, and attendance rates (Bhattacharya et al. 2006; Millimet et al. 2010; Frisvold 2015; Leos-Urbel et al. 2013). Evaluating a breakfast in classroom program (BIC), Imberman & Kugler (2014) found no effects on grades or attendance, however. Re-analyzing data from a randomized trial on a BIC program, Schanzenbach & Zaki (2014) report a few positive impacts on measures of dietary quality, and no positive impacts on behavior, health or achievement.<sup>11</sup>

The only paper on the long-term effects of serving a nutritious school breakfast is a recent paper by Bütikofer et al. (2016), where the long-term impact of a program that replaced the traditional hot meal at the end of the school day with a nutritious breakfast is evaluated. The results show a positive effect of the school breakfast on long-term economic outcomes. We differ from their paper in that we estimate the effect of introducing healthy school lunches whereas they study the impact of replacing a late meal with an earlier (and healthier) one. Moreover, their study takes place in the 1920s in Norway, where the meals were supposed to reduce malnutrition and increase caloric intake.

#### 2.3 Mechanisms

Nutritious school lunches in primary school can affect later life outcomes through multiple channels. First, the quality of the meals can have a direct impact on children's health and growth. Nutrition is believed to be the most important environmental factor affecting height and affects growth more in the postnatal period than in the prenatal period (Silventoinen, 2003). The single most important nutrient for height growth is protein, followed by minerals and vitamins A and D, and a recent study by Grasgruber et al. (2014) found that variation in the consumption of high-quality proteins from milk, pork, fish, and wheat explained most of the variation in height across 45 European countries. Vitamin D deficiency affects the mineralization of bones since vitamin D is important for the absorption of calcium, and this problem is more severe in northern regions due to a shorter light period in winter time (Silventoinen 2003). In addition, a number of studies have shown that calcium, phosphorus,

 $<sup>^{11}</sup>$ Kremer (2005) analyze a program in Kenya that randomized access to subsidized breakfasts in preschools. The results showed positive effects on school attendance and test scores.

magnesium, zinc and iron are of importance for human growth (Allen 1994; Prentice & Bates 1994).<sup>12</sup> Interestingly, the Swedish school lunch program aimed at increasing the intake of exactly these nutrients and in our empirical analysis we will therefore focus on height as an important marker for health and improved nutrition but also analyze other important health outcomes.

A poor diet can also lead to overweight and obesity with related problems such as high blood pressure, diabetes, heart disease, and risk of stroke (see for example Hoynes & Schanzenbach (2015)). While school meals can be a remedy for poor diet, they may also contribute to overweight and obesity, as discussed above, and in our analysis we will therefore study the effect of the Swedish school lunch program on measures of body size.

Second, poor nutrition in early life can directly affect the formation of cognitive skills and educational attainment. Deficiencies in zinc, iodine, iron and folate have been linked to worse cognitive development of school-aged children in developed countries (Pollitt & Gorman 1994; Delange 2000; Bryan et al. 2004; Feyrer et al. 2017).<sup>13</sup> School lunches may also affect educational attainment and test scores through making pupils more attentive and raising their energy level. Figlio & Winicki (2005) found that schools at risk of sanctions if students under-performed in tests and that served meals richer in carbs on the days of the tests improved their scores in several subjects, a finding that is also supported in experimental research (Wyon & Abrahamsson 1997).

Third, free school meals can affect outcomes through mechanisms not directly related to nutrition. The provision of free meals can make it more attractive for some students to attend school and lead pupils to attend more classes (Hinrichs 2010). Such a mechanism is unlikely to be of importance in our context since primary school was mandatory and since rates of non-attendance were low already before the program was introduced.<sup>14</sup>

Finally, the introduction of free school lunches can affect household income since food expenses are reduced and since more mothers were able to enter the labor market. The latter effect may have arised because mothers did no longer have to be at home and cook during the lunch break. To the extent that increases in income benefit the children in the household, as suggested in the study by Dahl & Lochner (2012), this represents another mechanism through which long-term outcomes may be affected. In our empirical analysis we will address this possibility by estimating the effect of the school lunch program within families. If increased household income is an important mechanism, we expect to see much smaller effects within families since all children should gain from such income increases.

 $<sup>^{12}</sup>$ While healthy school meals can offer a remedy through a more balanced diet, they could in principle also worsen things through a mismatch between early and late nutritional supply. If the supply of nutrients during the pre-natal or early post-natal is scarce, the body predicts that the future will also be nutritionally deprived and invokes biological mechanisms to adapt to the future environment (Barker 1997; Gluckman & Hanson 2005). This mechanism is referred to as the Barker hypothesis and predicts that when nutrients are plentiful in later period, but not in an earlier period, the early metabolic adaptions are a bad fit for the later environment and the risk of metabolic disorders increases.

 $<sup>^{13}</sup>$ See Gómez-Pinilla (2008) for a review on some of the biological mechanisms that link diet quality and nutrition to cognitive outcomes.

 $<sup>^{14}</sup>$ We show that this is the case in Section 7.7.4.

# 3 The Swedish school lunch reform

We next provide an overview of the Swedish school lunch program. In 1946, the Swedish social democratic government signed a proclamation, Författningssamling (1946), guaranteeing comprehensive state subsidies for all municipalities that introduced universal and free school lunches in primary school.<sup>15</sup> The motivation for this government program came from a number of studies on the nutritional standards of the food consumed by school children. Surveys showed that breakfast for most children in primary school comprised coffee, tea, or hot chocolate, in combination with white bread (SOU 1945). Only one third of the children received porridge or "välling" (a milk-based cereal hot drink). The surveys revealed that 50 percent of the children did not receive any hot meal (other than porridge, etc.) until the evening. There were also concerns about the fact that one quarter of the children were not able to eat their lunch at home because of long travelling distances and instead had to bring food prepared at home to their school. Researchers analyzing the nutritional content of the food brought from home, as well as the food consumed at home during lunch, concluded that there were serious problems with the content, which negatively affected almost all school children and, in particular, frail and sick children.<sup>16</sup> In particular, the intake of vitamins A, C and D, protein, and certain minerals was perceived as inadequate (SOU 1945). The concerns were only related to the quality of the meal whereas the quantities consumed were believed to be adequate.

At the start of the program, every municipality that informed the regional school board about their intent to serve school meals for free automatically became eligible for the subsidies as long as they complied with the recommended nutritional standards of the meals (Virgin 1970). The government program for providing free and nutritious school lunches included a provision of grants to cover both salary and ingredient expenses and, more important for our period of study, a large institutional "framework" aimed at helping and controlling municipalities to provide the best possible school lunch for the tax money being spent. While the coverage of costs for school lunches by government grants dropped from 70 percent in 1948 to 22 percent in 1955 (SOU 1958) the institutional provision of help and control had increased in a similar magnitude. In SOU (1958), a detailed description is given of this extensive program, hence, providing us with information on the actual treatment occuring.

Important for our study is that the school board was actively taking part in the work of providing lunches.<sup>17</sup> First, the school board collected statistics annually from each municipality on their costs or providing free lunches, on their organization, and their facilities. Second, they provided guidance on how to best organize these school lunches by actively visiting these schools and producing brochures including good examples, including what lunch

 $<sup>^{15}</sup>$ Some limited state subsidies were introduced already in 1937 but few municipalities applied for those, and the subsidies were targeted to the areas that were most severely hit by the economic crisis in the 1930s. School lunches were therefore rare before 1946, and in 1940/41, only 13% of primary school pupils were served a school lunch free of charge.

<sup>&</sup>lt;sup>16</sup>Similar concerns were expressed in the U.S. in the 1940s, where it was questioned whether children would be nourished adequately if they brought a lunch to school or ate lunch at home (Hinrichs 2010).

<sup>&</sup>lt;sup>17</sup>The school board (Skolöverstyrelsen in Swedish) was a central office for school issues, formed in 1920. The school board was originally responsible for primary schools and educational institutions, but subsequently, it also ruled over elementary and upper secondary schools. The school board was replaced in 1991 by the National Agency of Education.

ingredients to buy locally.<sup>18</sup> Third, in another brochure called "Kosthållet vid skolmåltider", they provided guidelines for the nutritional content of a typical meal being served, with a 3-week menu template being the main feature in these guidelines; see the template of a typical 3-week school lunch menu in Figure A1 in the appendix.<sup>19</sup> However, the nutritional standard of the meal was not for the municipality to decide, and hence, these 3-week menus were more than just guidelines.<sup>20</sup> Fourth, the school board realized that having such high requirements for the local authorities also required a lot of help in terms of providing education in these matters. These education programs involved several levels with initial education programs of kitchen staff lasting up to two months, while continuing education could be weeklong courses. The school board also frequently visited the schools, and at these visits, further education was given through discussion groups etc.<sup>21</sup> To conclude, the overall impact of this extensive government program resulted in practically all schools serving the so called A-meal, i.e., the one being presented in the 3-week menu templates, including a hot meal, sandwiches and milk, by 1955.<sup>22</sup> Hence, there is supporting evidence that practically all students living in a municipality having adopted the free school lunch program were served a nutritious school lunch.

Important for the interpretation of our results (intent to treat effects) is the school lunch participation rate. Early on in the program, schools were required to keep daily track of the number of pupils eating school lunches. However, for the later period of the program, pupil counts were no longer mandatory, since subsidies were much lower, but, among the municipalities that did report such figures, the school lunch participation rate was always above 90 percent. In addition, only approximately 4 percent of the pupils were absent from the school lunch when the program was evaluated in the late 1950s (SOU 1958). These participation rates are remarkably high considering that a low rate of school meal attendance is a common problem when studying the effect of school meals, making our empirical estimates close to being interpreted as treatment on the treated effects (Kristjansson et al. 2007; Bernstein et al. 2004; Evans & Harper 2009).

The government investigations on the quality of meals consumed before the school lunch program led to the conclusion that the school lunch should comprise freshly cooked hot food with an adequate amount of micronutrients, together with milk and bread. Detailed guidelines were provided by the National Medical Board regarding the amount of vitamins A, B, and C, protein, calcium, iron, phosphorus, and egg white, and the A-meal were to provide a third of the daily need of calories (>800 calories), compared to a caloric intake of approximately 500 calories if only eating two cheese sandwiches and drinking 3 dl of milk.<sup>23</sup>

 $<sup>^{18}</sup>$ Since salaries of kitchen staff amounted to approximately 1/3 of total costs improving the productivity of producing meals was important; see Table 6 in SOU (1958).

<sup>&</sup>lt;sup>19</sup>The content of the menus was decided in collaboration with the National Agencies for Medicine and Public Health.

 $<sup>^{20}\</sup>mathrm{On}$  page 25 in SOU (1958) it is stated that this brochure contains typical menus, which provide a benchmark for the standard of school meals.

 $<sup>^{21}</sup>$ To obtain some idea about the scale of these programs it is stated that each year 2-300 kitchen staff were given the education (pp. 24-26, SOU 1958).

 $<sup>^{22}</sup>$ Table 10 in SOU 1958 shows that 98% of the students were served an A-meal. The other 2 percent were served a B-meal (Oslo breakfast), while very few were served a C-meal (porridge).

<sup>&</sup>lt;sup>23</sup>The guidelines were as follows: protein: 32 grams. Calcium: 0.4 grams. Phosphorus: 0.8 grams. Iron: 7 grams. Vitamin A: 2000 I.E. Vitamin B: 0.5 mg. Vitamin C: 25 mg. Egg white: 65 grams. These amounts correspond to half of the average daily need of vitamins, minerals, and egg white for school-age children (SOU 1945).

These nutritional guidelines were implemented and expressed through the 3-week school lunch menus, often comprising meat-based stews, vegetable-based soups, fish and meat or egg-based dishes, and fruit, berry, or vegetable-based dishes (SOU 1945). With each lunch, 30 cl milk and rye bread with butter were to be served. Hence, one way to think about the change in the lunch at school before and after the reform is as adding a hot nutritious meal to the typical pre-reform milk and sandwich package. The guidelines also included recommendations about maximum fat content, illustrating an early awareness of the risks of being overweight.

Two features of the school lunch program are of particular importance for our empirical strategy. First, when school lunches were introduced in a municipality, it covered all pupils in all grades in primary school. During the time period considered, children started school in the year they turned seven and the number of years that a pupil was exposed to the program therefore depended on the grade the student was in when it was implemented. In our data, we therefore have cohorts that were never exposed to free school lunches, cohorts that received school lunches during their entire primary school period, and cohorts that were partly exposed for one or more years. Second, the school lunch program was introduced gradually across municipalities. In a government report from 1958, it was concluded that lack of proper facilities was the major bottleneck for municipalities, which partly explains the differences in the timing of the uptake (SOU 1958). In section 5, we conduct comparisons between early and late adopters of the school lunch programs to assess whether the different groups faced similar pre-treatment trends.

Finally, an additional motivation behind the introduction of school lunches was to ease the household work burden for women and to encourage females to enter the labor market. It was believed that the provision of school lunches would be an important policy in this regard, since women would no longer need to stay at home during daytime to cook lunch (SOU 1945). In the empirical analysis we address this potential mechanism.

#### 3.1 Sweden during the 1960s and other reforms

To put our results into context, some knowledge about Swedish society in the 1960s is useful. Sweden was not occupied during WWII, and the postwar period was a period of strong economic growth and a strong expansion of the welfare state. It is important to understand that hunger and malnutrition were not of great concern in the 1950s and 1960s, when the school lunch program was introduced. The fraction of children growing up in poor households during the 1960s was not very different from later rates and in 1968, for instance, approximately 18 percent of children were growing up in liquidity constrained households, which was similar to the rate in 2000 (Jonsson & Mood 2013). With a GDP per capita in 1960 of 11,871 USD (in PPP 2005 USD), Sweden belonged to the 5 richest countries in the world.<sup>24</sup> The environment in which the school lunches were served was thus very different from the situation in the 1920s and 1930s, when the limited school meal programs in place were targeted towards poor regions where malnutrition and food shortages were real concerns. During this time period, there were also no concerns about being overweight or about obesity in the Swedish society. From the military enlistment records, we know

 $<sup>^{24}</sup>$ The corresponding GDP per capital in the U.S. was 15,644 USD.

that only approximately 1 percent of men born during the 1950s and 1960s were classified as obese at age 18 (Neovius et al. 2008).

Another large reform that occured in Sweden during the 1950s and 1960s was the compulsory schooling reform. The reform increased the mandatory years of primary schooling from 7 to 9 years and was rolled out across municipalities between 1949 and 1969.<sup>25</sup> In our regressions, we include indicators of exposure to the schooling reform to not confound any effect of school lunches with those of the reform. Most municipalities had already introduced the schooling reform by 1959, and our results are not affected by controlling for reform exposure or by dropping municipalities that had not yet introduced the schooling reform. In addition to the schooling reform, there were no other reforms that were rolled out gradually across municipalities during the same time period.

### 4 Data

#### 4.1 Exposure data from archives

We collected and digitized information on the introduction of school lunches across municipalities from the Swedish National Archive. In the archive, the information was kept on paper forms, where each municipality, for each year, reported whether they served school meals, the number of pupils served, and their costs for the school meals. In January 1960, Sweden had 1031 municipalities, and we managed to collect data on 1004 of them.<sup>26</sup>

In our empirical analyses, we focus on the municipalities that introduced the school lunch programs between 1959 and 1969, in total 265 municipalities. The reason for focusing on this period is that the National Archive does not hold forms sent by the municipalities in earlier years. Before 1959, the records were kept by the National School Board ("Skolöverstyrelsen") but the records from this earlier period were thrown away more than 20 years ago.

In Figure 1, we illustrate how the school lunch program was introduced between 1959 and 1969. We see that the number of municipalities that introduced the program gradually increases during the period, with approximately 25 municipalities added per year on average. In Section 5 below, we conduct some comparisons on the characteristics of the early, mid, and late adopting municipalities and with municipalities that already had introduced the program by 1959. As it turns out, we find little evidence of any systematic patterns in the timing of the introduction of the program across municipalities.

In 1959, when our data series starts, approximately 750 municipalities had already introduced free school meals while 281 had not (of which we have information on 265). We lack information on which year these 750 municipalities introduced their school lunches but we do know that individuals in these municipalities who started school in fall 1959 or later must have been exposed to school lunches during their entire school period. Individuals born 1952 and onwards from these municipalities, who entered the first grade at age 7, are therefore included in our data set and empirical analyses. The treatment for those pupils is

 $<sup>^{25}\</sup>mathrm{See}$  Lundborg et al. (2014b) and Holmlund (2008) for a description of the reform.

 $<sup>^{26}</sup>$ Twenty-one municipalities lacked an introduction date as of 1969. Some of them are likely to have introduced school meals earlier but failed to report it properly.

equal to the number of years they spent in primary school, i.e., 7 or 9 years, depending on whether they were exposed to the 9-year compulsory schooling reform.

For the remaining municipalities that introduced the school meal program between 1959 and 1969, we include individuals born between 1942 and 1965. Their exposure to school meals differs according to which year they started school, which year the municipality started serving free school meals, and which year their municipality of residence introduced the compulsory school reform. Among these individuals, there will be individuals with between zero and nine years of exposure. Table 1 shows the distribution of school lunch exposure in our sample. Here, we see that 83 percent of our sample was exposed to school lunches for 9 years and 8 percent were never exposed. The fractions that were exposed between 1 and 8 years are much smaller since there will only be one cohort in each municipality that will be partially exposed for a certain number of years. For instance, only those who were in grade 6 when the school lunch program was introduced will have been exposed for 4 years. The fraction exposed for 7 years is larger, at 4 percent, reflecting that certain cohorts went through the "old" 7-year primary school.

#### 4.2 Register data and census data

In our empirical analyses, we employ data from a number of administrative registers that have been linked to census data and the historical data on school lunch exposure. Our starting point is the Register of the Total Population (RTB), where we select every Swedish citizen born between 1942 and 1965. We obtain information on municipality of residence during school age from the censuses in 1960 and 1965. For those born between 1942 and 1954, we use the 1960 census to determine their municipality. They will then be a maximum 18 years of age, and we know from previous studies that most individuals aged 18 and below lived in the same municipality as that in which they went to school (Holmlund 2008). For cohorts born after 1954, we obtain the information on municipality of residence from the 1965 census.

We obtain data on income from the income and taxation register (IoT). Our main income measure includes labor market earnings plus all taxable benefits such as unemployment benefits, sickness pay and welfare pay. The records start in 1968 and end in 2011 and are present at the yearly level. These records come from the equivalent of W2 records in the United States and are reported by employers to the tax authorities. To focus on long-term outcomes, we construct a measure of lifetime earnings, defined as the mean of yearly incomes between 1968 and 2011.<sup>27</sup> In our sensitivity analysis we will also test the robustness of our results to the exclusion of taxable benefits. The income measures are also used to create indicators of labor market participation of parents, defined as having any labor income a given year.

For data on educational outcomes, we use the education register (utbildningsregistret, UREG). These data are reported directly from educational institutions in Sweden and are available from 1990 and onwards. We use the latest available measure in the data and impute years of schooling from the highest obtained degree reported in the register. We also consider alternative education margins, such as transitioning into secondary and tertiary

 $<sup>^{27}</sup>$ The year 1968 is the earliest year for which we observe earnings data in the income register.

education. To measure parents' schooling, we use the 1970 census with self-reported level of education since the education register lacks reliable data for some of the older generations.

We obtain additional outcome measures for males from the Swedish military enlistment register. The register includes data on test results at the military enlistment, which occurs for males the year they turn 18. The register includes information from tests on cognitive skills as well as data from physical measurements, such as height and weight, and overall health assessments. The data exist for cohorts born 1951 and onwards, meaning that there are few cohorts that are entirely unexposed to the school lunch program and the analyses on this sample will therefore not include any municipality-specific pretreatment trends.

The cognitive ability test is a traditional IQ test, similar in style to the AFQT in the US. The test includes four sub-tests, Instructions, Synonyms, Metal Folding, and Technical Comprehension. The cognitive skill variable that we use in our analyses is a standardized version of the composite cognitive measure calculated by the military enlistment service. Previous studies have established that the measure of cognitive skills is a strong predictor of adult earnings (see, for example, Lundborg et al. (2014b)).

To measure the overall health of males we use data from the health examination at the military enlistment. Each male was given a score on a 13-step scale ranging from more or less perfect health, which is necessary for "high mobility positions" (such as light infantry or pilot) to very poor health, meaning that the individual is not allowed to undergo military training. The score is based on the health examination and on doctor certificates proving the existence of any health conditions. In our analyses, we use a binary indicator of being in more or less perfect health as an outcome. In some analyses, we also use data on diagnoses recorded at enlistment.

To study height, weight, and smoking behavior among women we use data from the Swedish Medical Birth Register, covering all births in Sweden. The same register is used to study outcomes of the children of the mothers, including the birth weight of babies. Note that the register only covers women in our sample that gave birth from 1973 and onwards, and we will in our empirical analysis therefore test how sensitive our results are to this potential source of sample selection.

Finally, we measure long-term health outcomes through data from the national hospital register and cause-of-death register. The former register covers all hospitalizations in Swedish hospitals together with diagnoses, and we use data from 1995 to 2013 while the latter covers all deaths, together with causes, from 1968 to 2013.

#### 4.3 Selection of data and descriptive statistics

Selecting individuals born between 1942 and 1965 and who were registered in the census of 1960 if born in 1954 or before and in the census of 1965 if born in 1955 or after, leaves us with a population of 2,177,167 individuals. Since we include individuals born between 1942 and 1965 in municipalities that introduced the school lunches in the period 1959-1969, and individuals born between 1951 and 1965 in municipalities that already had introduced the lunches in 1959, this further limits our sample to 1,073,451 individuals.

In our regressions, the number of observations varies somewhat across regression as the fraction of missing data varies across outcomes. In Table 2, we show descriptive statistics on the outcomes studied and on the background characteristics used in the regressions.

# 5 Empirical method

The timing of the implementation of the school lunch program across municipalities was not random and depended, as discussed above, on the availability of kitchens and staff. We address the non-randomness of reform exposure by exploiting a difference-in-difference (DiD) design. The identifying assumption is that conditional on municipality and birth cohort fixed effects, school lunch exposure is as good as random. If our identifying assumption holds, we obtain unbiased estimates of the effect of school lunch exposure on the outcomes studied.

We employ two different DiD specifications in our empirical analyses. Our first specification specifies the treatment variable as a linear years-of-exposure variable and can be written as:

$$y_{itm} = \beta_1 S_i + \beta_2 M_i + \beta_3 N_i + \beta_4 CSR_i + \lambda_t + \alpha_m + \epsilon_i \tag{1}$$

where  $y_{itm}$  is the outcome of interest, such as schooling or earnings of an individual *i* at time *t* in municipality *m*, *S* indicates the number of years for which the individual was exposed to school lunches, *M* is an indicator of being male, *N* an indicator of being born in Sweden,  $\lambda_t$  denotes birth cohort fixed effects, and  $\alpha_m$  denotes municipality fixed effects. Since the compulsory school reform that increased mandatory years of schooling from 7 to 9 was also implemented during the study period we also include a variable, *CSR*, indicating whether the individual was exposed to the compulsory schooling reform.

Our second specification is more flexible in that we include 8 dummy variables, one for each number of years of exposure. The reference group in this specification is those who were never exposed to school lunches in primary school. This specification allows for nonlinearities in the effect of treatment intensity and, in addition, it allows us to illustrate the point estimates graphically.

In our analyses, we also estimate models where we control for municipality-specific linear or quadratic time trends. For the trends to reflect pretreatment trends, we estimate them on a sample of individuals born between 1933 and 1941, who would have finished primary school before 1959, and who were thus not affected by the school lunch reform.<sup>28</sup> Using this sample, we run regressions on the different outcomes studied as a function of year of birth, municipality, and linear or quadratic municipality-specific trends. We then predict pre-reform trends for our main sample and include them as control variables in our DiD regressions.<sup>29</sup>

#### 5.1 Internal and external validity

An important assumption in any DiD-analysis is that pretreatment trends in the outcomes studied are parallel across treatment and control groups. We can provide direct evidence on the validity of this assumption by comparing pre-trends in outcomes between early, mid, and late adopting municipalities of the reform. At any point in time, the control

<sup>&</sup>lt;sup>28</sup>It is particularly important in our context that the trends reflect pre-treatment trends since it would be easy to confuse any gradually increasing effect for cohorts that were increasingly exposed to the reform with post-treatment municipality-specific trends. Recall that the oldest cohort exposed to the reform in a given municipality was exposed for only one year, the next-to-oldest for two years, etc.

 $<sup>^{29}</sup>$ For a discussion about this approach to control for trends, see, for example, Wolfers (2006), Holmlund (2008), and Lee & Solon (2011).

group will comprise individuals in municipalities that did not yet implement the reform, and it is therefore of importance that individuals in early, mid, and late adopting municipalities face similar pretreatment trends. In Panels A to C of Figure 2, we show pre and posttreatment cohort trends in adult income and education where we have divided the sample to individuals in early, mid, and late adopting municipalities. Those are defined as municipalities that introduced school lunches in 1959-1962, 1963-1965, and 1966-1969, respectively. In the same graphs, we plot the trends for municipalities that already introduced the reform by 1959 to shed light on the representativeness of the municipalities that introduced the reform between 1959 and 1969. The x-axis represents birth cohorts.

To measure income at the same age for all cohorts when constructing these plots, we focus on mean income during ages 35-45. The graphs in Panels A-C illustrate two important things. First, and focusing on the unexposed cohorts born prior to 1952, we see that the absolute levels of income and education are strikingly similar between early, mid, and late adopting municipalities and municipalities that already implemented the program. This suggests that there was no strong selection in the timing of the implementation of the reform with respect to these factors, reducing concerns about external validity. Second, the trends are similar across the four groups, suggesting that the parallel trend assumption is supported in our context.

In Panel D of Figure 2, we show trends in average municipality income for early, mid, and late adopting municipalities by *year*. Here, our data goes back to 1952, which means that we can study trends in municipality characteristics for 7 years before the first cohort was exposed in our sample. Trends and absolute levels are strikingly similar when we study average income per capita.<sup>30</sup>

In the robustness section below, we provide further evidence on the validity of the parallel trend assumption by performing an event study analysis on the effect of reform exposure on earnings.<sup>31</sup> Morever, we test for a number of additional threats to our DiD design. One such threat would be parents moving with their children across municipalities in response to the introduction of the lunch program, and in the robustness section, we check for any evidence of systematic moving patterns.

# 6 Results

We begin our empirical analysis by estimating the long-term economic impact of the school lunch program. We then turn to potential mechanisms in Section 7 and to intergenerational effects in Section 8.

#### 6.1 Lifetime income

Table 3 shows estimates of the long-term effect of the school lunch program on adult income. Panel A shows results from the specification that enters the number of years of school lunch exposure linearly whereas the specification in Panel B instead uses a set

 $<sup>^{30}</sup>$ We have also looked at trends in average real estate value, and the findings are similar (available on request).

 $<sup>^{31}</sup>$ We have also run placebo-like DiD regressions where we regress the number of years of school lunch exposure on parents' years of schooling. Taking the average of the parents' schooling in the census of 1970, which is the earliest year for which we have data on parental schooling, we obtain a small and insignificant estimate of -0.00008 for the effect of parents' schooling on their children's number of years of school lunch exposure. These results are available on request.

of dummies to capture any non-linearities in the effects. Such non-linearities can arise, for instance, if the effect of school lunch exposure strongly depends on at what age the exposure starts. Recall, however, that we cannot distinguish the effect of treatment intensity from that of age at first exposure to school lunches, as earlier exposure always means greater cumulative exposure to school lunches. This arises from the feature of the program where the treatment stays on once it starts.<sup>32</sup> All specifications control for municipality fixed effects, cohort fixed effects, whether born in Sweden, gender, and schooling reform exposure. In addition, we show specifications that include municipality-specific linear and quadratic trends (columns 2 and 3).

The results in column 1 of Panel A show that one additional year of exposure to the school lunch program increases adult income by 0.35 percent. If we extrapolate this effect to 9 years of exposure, we obtain an effect of approximately 3 percent for those who were exposed during their entire primary school period. The estimate is robust to controlling for linear and quadratic municipality specific trends, as shown in columns 2 and 3.

We next run regressions where we include a full set of dummy variables for each number of years of exposure, with zero years being the reference category. The results are shown in Panel B, where the dummy indicating 9 years of exposure is significant and positive, suggesting that exposure during the entire primary school period is associated with almost 3 percent higher adult income. This estimate comes close to the implied effect of 9 years of exposure that we obtained in Panel A. Exposure to school lunches for between 6 to 8 years also significantly raise income, but the effects decline in magnitude and significance as the number of years of exposure decline. For those who were exposed only between 1 and 5 years, the estimates are insignificant and much smaller in magnitude. The pattern in the point estimates suggests a clear dose-response relationship where greater exposure is related to greater effects, consistent with previous findings where earlier exposure to a positive intervention have greater effects than later exposure (van den Berg et al. 2014; Chetty et al. 2016).

To illustrate the dose-response relationship visually, we plot the point estimates in Figure 3. Here, the point estimates are close to zero for those exposed for between 1 and 3 years, corresponding to a onset of treatment at ages 13-15. The estimates start increasing thereafter but the largest effects are seen for those exposed for 7 to 9 years, corresponding to treatment starting at ages 7 to 9.

While the estimates in Table 3 are intention-to-treat estimates, they likely come close to average treatment effects, as the participation rate in the school lunch program was above 90 percent. The high participation rate of the Swedish school lunch program is an interesting feature of the reform and makes it different from other programs, such as the U.S. food stamp program, where the participation rate was approximately 43 percent (Hoynes et al. 2016).<sup>33</sup>

The increase in earnings by 3 percent for those fully exposed to the program can be regarded as a substantial effect. We can relate it to estimates of the returns to schooling in

 $<sup>^{32}</sup>$ The school lunch program shares this feature with other social programs, such as the food stamp program and the Moving to Opportunity Program (Hoynes et al., 2016; Chetty et al., 2016).

<sup>&</sup>lt;sup>33</sup>Since we lack data on individual participation in the program, we are unable to provide instrumental variables estimates of the effect of participation in the school lunch program on earnings.

Sweden (in terms of lifetime income). Studies by Black et al. (2015) and Lundborg et al. (2014a), for instance, provide such estimates, where the effect of one additional year of schooling amounted to 4-5 percent greater earnings. The effect of 9 years of school lunch exposure on earnings thus corresponds to the effect of an additional two thirds of a year of schooling.

Since there are no other studies that estimate the long-term effect of access to school lunches on adult income in developed countries, it is difficult to compare our findings to other studies. The size of the effect is in the same ballpark, however, to the estimated effect of access to school breakfast in Norway, where the long-term effect on earnings was estimated to 2-4 percent (Bütikofer et al. 2016). We can also relate our finding to other school-based interventions; Fredriksson et al. (2013), for instance, find that a one-pupil reduction in class size in Swedish schools increases adult earnings by 1.2 percent, suggesting that the effect of school lunches corresponds to that of reducing class size by 2.5 pupils.

#### 6.2 Timing of school lunch exposure

As explained above, our estimates capture the combined influences of length of exposure to school lunches and age at onset. An alternative specification of our empirical model is to nonparametrically estimate the effect of age at onset such as done in Hoynes et al. (2016). For this purpose, we re-run our main DiD-specification and include dummies indicating the individual's age at which school lunches were introduced. The omitted reference category is being above the school leaving age when school lunches were introduced in one's municipality.

Figure 4 plots the estimates from the age-at-onset regression. The effects drastically declines when moving beyond age 9, and it appears critical to receive the school lunch at ages 7-9, i.e., an early onset. Beyond these ages, there is a minimal effect of obtaining access to school lunches. The effects are thus similar to what we found above, where the greatest effects were obtained for those exposed to school lunches for 7 to 9 years. This is unsurprising since the age-at-onset effects are almost a mirror of the length-of-exposure effects.<sup>34</sup> Although we are unable to distinguish the effect of age at onset from that of length of exposure, we are still able to conclude from our results that the school lunch program produces larger benefits for children who were young when first exposed.

#### 6.3 Heterogeneity by parental income

So far, we have focused on the mean impact of the school lunch program and we next move on to examine whether some pupils were affected more than others. Policy-makers at the time were particularly concerned about the nutritional intake of poor and sick pupils, and we therefore next investigate whether there is important heterogeneity in the effect of the program by parental income. Moreover, we investigate whether there are differential effects by gender since some previous studies suggest gender differences in the effect of school lunch programs (Hinrichs 2010).

In columns 1 to 4 of Table 4, we show the effect of school meal exposure on earnings

 $<sup>^{34}</sup>$ They do not exactly mirror each other, however, since individuals with the same number of years of exposure can have different age-at-onsets in the new and old school systems. A person who has 7 years of exposure in the 9-year school was exposed to school lunches from grade 3 and onwards whereas 7 years of exposure in the 7-year school system meant having school lunches from grade 1 and onwards.

by parental (mean household) income quartiles.<sup>35</sup> Children of parents who belonged to the lowest quartile of the income distribution (column 1) were affected to a greater extent than other children. The point estimate, 0.0057, is more than double in size compared to the estimates obtained for children whose parents belonged to the second and third quartiles of the income distribution. The lowest quartile also maps onto the definition of relative income poverty, amounting to having a disposable income less than 50% (OECD) or 60% (EU) of the median (Jonsson & Mood 2013).<sup>36</sup> At the fourth quartile of the parental income distribution, the point estimate is much smaller and statistically insignificant. Children from low-income families thus seemed to gain more from the school lunch program, which may reflect that increase in school lunch quality was greatest for the poorest children.<sup>37</sup>

In columns 5 and 6, we turn to results by gender and find that the effect of school lunches on earnings is similar across men and women.

# 7 Mechanisms

We go on to examine the effect of the school lunch program on a number of intermediate outcomes, with the aim of shedding light on potential mechanisms behind our estimated income effects. We first focus on short-term outcomes measured for both males and females including height, body mass index (BMI), and overweight. For males, these outcomes are measured at age 18 during the military enlistment for cohorts born between 1951 and 1960, as no digitized records exist for earlier cohorts. For females, we can study height and obesity through information taken from the medical birth register from 1973 and onwards. In addition, we observe the outcome of cognitive skills tests and health assessments for males but not for females.

The data obtained from the military enlistment and from the medical birth register restrict our analysis in certain ways. First, we cannot include municipality-specific trends in the regressions since almost all cohorts are exposed to some extent and since we cannot predict pre-reform trends using unexposed cohorts. We take some comfort in the fact that our estimated effects of the school lunch program on income were insensitive to the addition of linear and quadratic municipality-specific trends. Second, we only observe height and weight for females who had a birth in 1973 or later. We thus lose those who had all their births prior to 1973 and those who never had children and, below, we therefore analyze to what extent school lunch exposure affects the likelihood of being observed in the medical birth register.

#### 7.1 Height, BMI, and overweight/obesity

Table 5 shows estimates of the effect of years of exposure to school lunches on height, BMI, and overweight/obesity.<sup>38</sup> The latter two outcomes are motivated by U.S. studies,

 $<sup>^{35}</sup>$ We measure parental income in 1968 when approximately 70 percent of the children were still in school.  $^{36}$ The lower threshold using 50% of the median is 85,000 SEK, while the larger threshold (60%) is 102,000 SEK. The threshold for the lowest quartile in our data is 93,000 SEK. Using the 60% threshold we obtain a significant point estimate of 0.0063.

 $<sup>3^{7}</sup>$ We obtain less strong evidence of heterogeneity in the effects by parental education. The point estimate for those with low-educated parents (both parents having only primary school education) is similar to that of those with more educated parents, although it is significant only in the former case. The similarity can be explained by the relatively compressed wage distribution for the parent generation in our data.

<sup>&</sup>lt;sup>38</sup>Those with a BMI above 25 are defined as overweight or obese.

where the results suggest that exposure to school lunches increases the probability of being obese (Millimet et al. 2010; Schanzenbach 2009). Including these outcomes also allows us to measure whether the school lunch program affected the amount of calories consumed, in addition to nutrition.

In column 1 of Table 5, we see that school lunch exposure has a positive and significant effect on male's height where one additional year of exposure increases height by 0.07 cm. If we extrapolate this effect to 9 years of exposure, we obtain an implied increase in height of approximately 0.6 centimeters. This is a large effect since it corresponds to 6-7 years of secular height growth in Sweden during the 1940s and 1950s (Werner 2007). We can also relate this finding to estimates of the (male) height premium in Sweden, where a 0.6 centimeter growth in height would imply 0.4 percent greater earnings, suggesting that the increased height following the school lunch program could explain approximately 15 percent of the adult income effect (Lundborg et al. 2014b).

In Panel B, we see some evidence of non-linearities in the height effect, where the magnitude jumps and becomes statistically significant at the 5 percent level for those with 4 or more years of exposure. In magnitude, the greatest effect is obtained for those with 9 years of exposure, where the effect is to increase height by 0.8 cm. In Panel A of Figure 5, we illustrate these patterns graphically.

In column 2, we show the results for females based on height data from the birth records.<sup>39</sup> Similar in magnitude to the effect among males, each year of school lunch exposure adds 0.05 cm of height. We see some evidence of non-linearities in Panel B, also illustrated in Panel B of Figure 5, where those exposed for 9 years gain 0.65 cm of height. This is a somewhat smaller effect than the one we obtained for males, but the relative effect is larger given that females are shorter on average.

We can relate the findings for height to the content of the school lunch program, where an important aim of the lunches was to increase protein and vitamin intake. Protein is believed to be an important input in height growth, and a recent study relates the variation in height across European countries largely to variation in protein intake (Grasgruber et al. 2014).

In columns 3 to 6, we show the effect of school lunches on BMI and overweight/obesity, defined as having a BMI above 25. The results do not provide any evidence that the Swedish school lunch program increased the risk of being overweight or obese, and the point estimates are in most cases close to zero.<sup>40</sup> Moreover, we obtain no evidence of any non-linearities in the effect, as demonstrated in Panels C to F in Figure 5.<sup>41</sup>

Our results on body size differ from results obtained in several U.S-based studies, and one likely reason for the divergence in results is that the Swedish program followed strict

 $<sup>^{39}</sup>$ For females who we observe several times in the birth register, we take the average of their height measures, and each woman appears only once in the regressions. Since birth records on height are available from 1973 and onwards, we observe height for only 30 percent of the female sample. When we a run regression on the effect of school lunch exposure on the probability of observing height we obtain a small but significant and positive effect of 0.002, suggesting that 9 years of exposure increase the chance of observing height by 2 percentage points from a baseline of 30 percent. We should therefore view the height results for females with some caution.

 $<sup>^{40}</sup>$ For women who are observed multiple times in the medical birth register we use their average BMI as an outcome.

 $<sup>^{41}</sup>$ We have also studied the effect of school lunches on the probability of being underweight (BMI<18.5). We find small and insignificant effects (results available on request).

guidelines regarding nutritional content and fat content.

#### 7.2 Health assessment at military enlistment

We next study the effect of school lunches on the assessment of overall health for males, measured during the military enlistment. The outcome is a dummy indicating having near perfect health in relation to typical tasks performed in the army. The results in column 1 of Table 6 show that school lunch exposure improves health, where those exposed for 9 years have a 4 percentage points greater likelihood of being assessed in near perfect health, corresponding to a 6 percent increase at the mean. In Panel A of Figure 6, we see that the shape of the relationship between school lunch exposure and health resembles the corresponding one between school lunch exposure and height, providing additional support that nutritious school lunches had positive effects on health, at least in the short run.

The military also records specific diagnoses at the enlistment, although there is limited variation in the data since 18-year-old males are generally in good health and since the unhealthiest males do not enlist (Carlsson et al. 2015). In Table A1 in the appendix, we study some diagnoses that could potentially be linked to improved nutrition and that may map into the overall health assessment: Type II diabetes, conditions of the digestive system, and visual impairment. While the incidence of diabetes is below 1 percent in the sample, 10 percent are classified as having a condition of the digestive system and 2 percent as having a visual impairment. Moreover, we study the effect of the probability of having any diagnosis and on the probability of having a congenital disorder (birth defect), where the latter should not be affected and thus acts as a placebo outcome.

In column 1 of Table A1, we see that school lunch exposure decreases the likelihood of having any diagnosis. We also find that school lunches decrease the risk of being diagnosed with diabetes at age 18, although it should be noted that the estimate is significant only at the 10 percent level. For digestive conditions and visual impairment, we also find beneficial effects of school lunches, and this time the estimates are significant at the 1 percent level. Finally, as our placebo test, column 6 reports that school lunches had no effect on an outcome that should not be affected: congenital disorders.

#### 7.3 Cognitive skills

In column 2 of Table 6, we show results for cognitive skills among males, as measured through the combined score of the cognitive ability tests at military enlistment. There is a positive estimate for years of exposure to school lunches, but it is imprecisely estimated. Panel B, or even clearer in Panel B of Figure 6, also reveals no sign of any positive relationship between years of exposure and cognitive skills. We have also elaborated with splitting the cognitive test score measure into two different types of intelligence, crystallized and fluid intelligence, as defined in Carlsson et al. (2015), without a change in results.

An absence of an effect on cognitive skills might reflect that the school lunch policy was introduced at ages beyond critical development periods for cognitive skills. A large literature suggests that cognitive skills are developed early in life, although the exact critical ages are a matter of debate Heckman (2007).

#### 7.4 Behavioral changes

Another question of interest is whether the school lunch program led to any permanent behavioral changes, such as adopting a healthy lifestyle. As a marker for this, we examine the smoking behavior of mothers, recorded in the medical birth register.<sup>42</sup> Admittedly, smoking does not reflect all aspects of lifestyle, and we lack data on other important markers such as physical exercise, drinking, and adult eating habits. As shown in column 3 of Table 6, we obtain no evidence that school lunches led to any changes in lifestyle, as the effect on smoking was small and insignificant. We analyze another potential behavioral outcome in Section 8, where we look for effects across generations.

#### 7.5 Education

We next turn to educational attainment as a possible mechanism through which exposure to school lunches can lead to higher adult earnings. In column 4 of Table 6, we focus on years of schooling but in Table A2 in the appendix, we also show results for entering university. Since education data are available for our entire sample, and not only for the military enlistment sample, we can study the effects for all cohorts between 1942 and 1965, and for both males and females.

We see that school lunch exposure positively affects schooling, where the effect of one additional year of exposure is to increase years of schooling by 0.03. In Panel B, 9 years of exposure increase years of schooling by 0.28. The estimates are illustrated graphically in Panel C of Figure 6. Given estimates of the returns to schooling in Sweden of 4-5 percent, an effect of 0.28 can explain almost half of the effect of school meal exposure on life-time income.

In column 1 of Table A2 in the appendix, we see that exposure to school lunches also increased the propensity to enter university. Those exposed for 9 years have a 1.5 percentage points greater likelihood of entering university, and the estimates are illustrated in Panel D of Figure  $6.^{43}$ 

#### 7.6 Long-term health outcomes

We continue by analyzing whether exposure to the school lunch program had long-term effects on health, as measured through adult morbidity and mortality. In column 1 of Table A3 in the appendix, we see that school lunch exposure had only a small and insignificant effect on mortality, measured through the probability of not surviving until 2013, which is the last year for which we have data from the mortality register. The finding is not surprising since the average age in 2013 is 56 in our sample and only 5 percent had died by then. As a measure of morbidity, we examine whether the individual had been hospitalized at any point between 1995 and 2013. The results, shown in column 2 of Table A3, again reveal

 $<sup>^{42}</sup>$ The outcome variable takes on the value 1 if the mother is observed to smoke at any time of measurement in the birth register. Smoking status is observed for 32 percent of our women and, again, school lunch exposure is associated with a slight increase in the likelihood of being observed.

 $<sup>^{43}</sup>$ In columns 2 to 3 of Appendix Table A2, we show the results for years of schooling and university attendance with controls for municipality-specific trends included. This does not change the results to any important extent.

small and insignificant effects of school lunch exposure. Since we lack data on outpatient care, we are unable to examine less severe health outcomes that do not require hospital care.

In addition, we have analyzed the effects on cause-specific morbidity where we focus on broad categories of diseases that could potentially be linked to school lunches such as cardiovascular disease, cancer, and respiratory disease. As shown in columns 3-6 of Table A3, we do not obtain any significant effects, suggesting that the school lunches did not affect the likelihood of experiencing severe health shocks during the period from young adulthood to middle age.<sup>44</sup>

#### 7.7 Alternative mechanisms

As discussed above, introducing free school lunches can affect the income of the households in two ways. First, since the school lunches were provided free of charge, their introduction meant that parents had to spend less money providing lunches for their children. If some of the money was instead spent on inputs that affected the health and human capital of their children, this may explain part of the positive effect of school lunches. Second, an important goal of the school lunch program was to free women from the task of providing lunches to their school-aged children and thereby increase female labor market participation. If the labor supply effect was substantial, increased household income would be one possible mechanism through which the introduction of school lunches affected school children. Dahl & Lochner (2012), for example, found that income increases among poor families in the U.S. increased their children's math and reading scores. In the following section, we provide evidence on the potential importance of these alternative mechanisms.

#### 7.7.1 How much did households save from free school lunches?

One way to understand the magnitude of any increases in household income coming from reduced food expenditures is to calculate how much a typical household would save. We know from the data collected exactly how much the schools did spend per student. If we assume that the cost per pupil represents an upper bound for what the meals that were previously served at home, or brought to school, would cost, we can make some rough calculations about potential cost savings. The yearly average spending per pupil was 3,730 SEK, which corresponds to approximately 1% of the average income of an household in 1968 (272,000 SEK).<sup>45</sup> We can assess whether a 1 percent increase in household income per child could generate the effect of household income on child outcomes. Dahl & Lochner (2012) used changes in the earned income tax credit program in the U.S. to estimate the effect of family income on child outcomes. They found that a \$1000 increase in household income, corresponding to a 3 percent increase at the mean, leads to a 6 percent of a standard deviation increase in math and reading test scores. If we would translate these findings to the Swedish context, where the increase in household income at most amounted to one

 $<sup>^{44}\</sup>mathrm{We}$  have also analysed the effect on cause-specific morbidity by parental income but find no evidence of heterogeneity.

 $<sup>^{45}3,730</sup>$  SEK corresponds to approximately 420 USD (in 2017).

percent per child, it is doubtful that the savings at the household level would explain any important part of the effects of the school lunch program that we estimate.

#### 7.7.2 Did the free lunches affect maternal labor supply?

We next estimate the effect of the school lunch program on mothers' labor supply in 1970, measured as having any labor income in that year.<sup>46</sup> The results in the fifth column of Table 6 show that having a child fully exposed to the school lunch program increases maternal labor market participation by 2.8 percentage points from a baseline rate of 53 percent.<sup>47</sup> This result shows that one of the aims of the reform, to increase maternal labor supply, was accomplished to some extent.

Among fathers, labor market participation was already high, and since mothers were usually the ones providing lunches at home, we expect less of an effect among fathers. As a placebo test, we therefore run the same analysis on male labor supply where we indeed obtain small and insignificant estimates; see sixth column of Table 6.

#### 7.7.3 Effects within families

Another way to shed some light on the role that increased household income, either through the effect of reduced household expenditures on food or through increased maternal labor supply, play for our estimated school lunch effects is to look at the effect of school lunch exposure within families. That is, we add a mother fixed-effect to our main DiD regression model, meaning that we identify the effect of differential exposure to school lunch exposure across siblings. If the positive effect of school lunches is explained by a positive household income effect, other children in the family might benefit as well.<sup>48</sup> If such an income increase was evenly distributed across the children, all children may benefit to at least some extent, and we would expect a smaller, or non-existent, school lunch effect within families. A smaller effect would also result if there were strong (positive) spillover effects between exposed and unexposed siblings, i.e., if the SUTVA assumption was violated.<sup>49</sup>

The estimate from models that add family fixed effects would still reflect increased household income if older siblings who were not exposed to school lunches had moved out of the household and, thus, were not affected by the income increase. Moreover, siblings unexposed to school lunches but who remained in the household would in any case be exposed to any income increase at the household level for a shorter period of time compared to their younger exposed siblings. To reduce these concerns, we run our family fixed regressions on a sample of closely spaced siblings, born at most 5 years apart.<sup>50</sup> In column 1 of Table 7, we first run our baseline DiD specification on families where the first and second born siblings are born at most 5 years apart in order to assess the external validity of the siblings

<sup>&</sup>lt;sup>46</sup>Approximately 70 percent of the children in our sample were still in school in 1970.

 $<sup>^{47}</sup>$ The results do not change if we add municipality specific trends, as shown in column 4 of Table A2 in the appendix.

<sup>&</sup>lt;sup>48</sup>A positive household income effect would also arise if local farmers gain from the school lunch program because of an increase in the demand for their products.

<sup>&</sup>lt;sup>49</sup>Alternatively, there could be knowledge spillover effects in meal provision, so that parents also start serving healthier meals. In such a case, we would expect that siblings unexposed to the school lunch program would also gain.

 $<sup>^{50}</sup>$ In these regressions we include families of any size but only use the first and second born children in the estimation. The average birth spacing in this sample is 3.7 years.

sample. The results are similar to our main findings in Table 3. When we add family fixed effects, shown in column 2, the point estimate is similar in magnitude and remains positive and significant. We obtain a similar point estimate when we run the analysis on second and third born children, as shown in column 3, but it is somewhat less precisely measured.

In column 4, we study male height as an outcome using within family variation and restricting the sample to first and second born brothers born at most 5 years apart. The point estimate is similar to the one we obtained without family fixed effects in Table 4 but the estimate is insignificant due to the much smaller sample size. For females, the sample size shrinks substantially, resulting in an insignificant point estimate of 0.033, which is not very different in magnitude from the corresponding estimate we obtained without fixed effects, however. In column 6, we show that the estimate of school lunch exposure on health assessed at military enlistment increases in magnitude when we add family fixed effects and remains significant at the 5 percent level.

Taken together, these results do not suggest that our main estimates are driven by income increases at the household level following from the introduction of school lunches. Any income gains in terms of reduced expenditures on food or from increases in labor supply should benefit all children in the household, at least for closely spaced siblings. The fact that the family fixed estimates are similar to our main estimates also implies that parental reinforcing or compensating behavior appears less important in our context. This differs from Almond et al. (2009), for instance, who found evidence of reinforcing behavior when studying the effect of radioactive fallout in Sweden on child outcomes. One reason for the difference may be that parents in our case were only able to adjust their investments in their children from age 7 and onwards whereas the parents in Almond et al. were able to react to child outcomes already from the birth of the child. If there are critical periods for parental investments, this might thus explain the difference.

#### 7.7.4 Did the lunches affect school attendance?

If children were more likely to attend school after the introduction of free school lunches, this may constitute one mechanism through which long-term income was affected. We consider this to be an unlikely mechanism in our context, as primary school was compulsory and since non-attendance rates were remarkably low in Sweden during the time period studied. Cattan et al. (2017) report that for cohorts born between 1930 and 1935, i.e., before the school lunches were introduced, students in grade 1 missed only 1.7 days on average for reasons other than sickness. The corresponding number in grade 4 was 3.3 days.

Truancy rates were also low in later periods (Jönsson, 1990). In a survey from 1967, truancy was almost non-existent for pupils attending 3rd grade, while increasing to 1 percent in 6th grade and to almost 4 percent in 8th grade. A similar rate, a truancy rate of 3 percent for 6th graders, was found in a study conducted in 1953, where many schools had not yet introduced school lunches. Another study conducted in 1968 found that truancy rates in 8th grade were 3.6 percent for boys and 1.6 percent for girls, see Jönsson (1990) for more details.

The low rates are not surprising given that the Swedish compulsory schooling law strongly prohibits truancy, and if children do not comply, fines will be given to their parents. Moreover, there was a strong focus on order and conduct in the school system at the time, and at the end of the school year, days of absence for different reasons (e.g., sickness, truancy etc.) during the school year were presented along with the grades in different subjects. It thus appears unlikely that increased attendance plays an important role for our results.

# 8 Effects across generations

We next analyze whether the school lunch program had effects across generations. Medical evidence suggests that inadequate intake of vitamin D can cause adverse pregnancy outcomes including intrauterine growth restriction and neonatal low birth weight (Pérez-López et al. 2015), and if greater exposure to school lunches meant that food habits were affected permanently, we may see an effect on birth outcomes. For this analysis, we use data from the medical birth register, where we observe the birth weight of children born to mothers in our main sample.<sup>51</sup>

Table 8 shows the results on child birth weight. In column 1, we see that children whose mothers were more exposed to the school lunch program were not different from other children in terms of birth weight. When we focus on indicators of having at least one child born with low or very low birth weight, we obtain negative but insignificant point estimates (column 2 and 3).

# 9 Robustness analyses

In this section, we present the results of several sensitivity checks. First, we perform an event-study analysis, where we look for the existence of pre-trends in the data that would violate our parallel trend assumption. Second, we test for strategic relocation across municipalities in response to the school lunch program. Third, we check whether our results are robust to the income measure we use and to dropping those that were exposed to the old 7-year primary school system. Finally, we investigate whether our estimated income effects are also obtained on our restricted samples from the military enlistment register and from the medical birth register.

#### 9.1 Event-study results

The key identifying assumption in our DiD specification is that of parallel trends. To shed further light on the validity of this assumption, we next perform an event-study analysis where we use the same specification as in our main analysis but add a number of dummy variables representing cohorts that were not exposed to the school lunch program. If the assumption of parallel trends is valid, we expect that the coefficients of these pretreatment dummies are zero.

 $<sup>^{51}</sup>$ For mothers with multiple births, we take the average of the children's birth weight in our regressions so that each mother appears only once in the data. When we study indicators of low birth weight (<2,500 grams) and very low birth weight (<1,500 grams), we use indicators of observing at least one child with low or very low birth weight. We observe the birth weight of at least one child for 40 percent of our women. A (DiD) regression on the likelihood of observing birth weight as a function of school lunch exposure produces a small and insignificant estimate of 0.0009.

We plot the point estimates from the event study regression in Figure 7. In the graph, years of exposure coincide with cohorts. Zero years of exposure correspond to the cohort that had just left school when school lunches were introduced. Minus 1 corresponds to those that left school more than 1 year before the school lunches were introduced, etc. Negative event time thus reflects the cohort distance in years from the first cohort treated with school lunches. In the graph, 1 year of school lunch exposure represents the oldest cohort exposed to school lunches, i.e., those who began the last grade when school lunches were introduced and who, thus, were only exposed for one year. Positive years of exposure thus represent both the intensity of school lunch exposure and the distance in years from the first cohort that was treated with school lunches.

The pattern in the figure shows no evidence of any pre-trends; conditional on common cohort effects and municipality effects, the point estimates of the pre-treatment dummies are all close to zero and insignificant. After treatment sets in, however, we see a gradual increase in the estimates, reflecting the gradual increase in treatment intensity for post-treatment cohorts. These results, together with the ones reported in Section 5, provide evidence on the validity of the parallel trend assumption.

#### 9.2 Strategic relocation

If some parents decide to move with their children in response to the introduction of school lunches in a municipality, our DiD estimates might be biased. Since most individuals in our sample were registered in both the censuses of 1960 and 1965, we can check whether they moved as children and whether such moves influenced their school meal exposure. In particular, we are interested in whether certain types of families moved, and for that purpose, we regress a dummy for moving to, and then a dummy for moving from, a municipality that had introduced the school meal program as a function of parental education and birth year. The results suggest that years of parental schooling significantly affect the probability of moving both to and from a municipality that introduced school lunches. The probability of moving to a municipality that would increase the total number of years that the person had school meals was 0.2 percentage points higher for someone whose parents had the maximum amount of schooling in the sample compared to someone whose parents had the least. The corresponding effect of moving to a municipality that would decrease the number of years of exposure was 0.4 percentage points. The coefficients are in both cases thus minute and the effect works in both directions. Rather than strategic relocations, we therefore conclude that these effects reflect the fact that people with higher education were in general somewhat more mobile.

# 9.3 Alternative income measures and dropping those in the 7-year school system

In our main analysis, we use a measure of income that includes labor income and all other taxable benefits, such as sickness insurance benefits and unemployment benefits. We next test whether our income results are robust to an alternative measure of income that excludes taxable benefits. This measure captures the "pure" labor income and does not include any compensation paid by the social welfare system, such as unemployment benefits or sickness benefits.<sup>52</sup> If school lunches make some people healthier and more productive in the long run, and thereby reduce sickness insurance take-up, such an effect would to some extent be hidden when we use an income measure that includes sickness insurance benefits. As seen in column 1 of Table 9, the effect of the school lunch program is almost unchanged when we use this alternative measure. This result is in line with our previous results of no long-term health effects of school lunches.

In column 2, we instead use average earnings during ages 35-45 as an outcome, allowing us to measure earnings at the same ages for all cohorts. The effects of school lunch exposure is still positive and significant at the 1 percent level, although the point estimate is smaller in magnitude, suggesting that the full benefit of school lunches has not yet fully materialized at these ages.

In column 3, we show regressions where we drop those that were exposed to the "old" compulsory schooling system with only 7 years of schooling as mandatory. This leaves our estimate of the effect of school lunches practically unchanged.

# 9.4 Income effects for those in the enlistment register and medical birth register

In our analyses on potential mechanisms, we used data from the military enlistment register and from the medical birth register. We next examine whether the positive effect of school lunch exposure on income that we observed above also exists for the individuals observed in these registers. For the analysis on males, we use register data on all males born between 1951 and 1965. The reason is that we do not have linked income data in the military enlistment register. We have enlistment data for 87 percent of this sample, which explains the smaller number of observations in the regressions on enlistment outcomes. While we cannot observe in the population income register which males actually enlisted, column 4 shows that we obtain similar point estimates for the male cohorts born between 1951 and 1965 as well as for the cohorts born between 1942 and 1965, which we used in our main analysis. We also obtain similar income effects for women for whom we observe height and the birth weight of their children in the medical birth register; see columns 5 and 6.

# 10 Did the benefits outweigh the costs of the program?

The Swedish school lunch program was extensive, and an important question is whether the benefits of the program exceeded its costs. Since we have detailed data on the costs of the program, we are in a good position to conduct such an analysis.

Our estimates suggest that exposing a child to 9 years of healthy school meals will increase the child's lifetime earnings by approximately SEK 102,000 (\$11,700), discounting future earnings at a 3-percent interest rate and counting earnings from age 21 to 65. This is almost four times the total discounted cost of the program, which amounts to SEK 26,900 (\$3,080). The benefit-cost ratio increases substantially if we single in on the children in poor families (bottom quartile of household income). For these children, 9 years of exposure increased earnings by 5.45 percent, meaning that the discounted benefits were seven times

 $<sup>^{52}\</sup>mathrm{Moreover},$  it does not include income from self-employment.

larger than the discounted costs. One should keep in mind that this simple back-of-theenvelope calculation only counts the income benefits of the school lunch program and that other important benefits are not accounted for.

# 11 Concluding remarks

In this paper, we provide estimates of the long-term economic benefits of a program that offered free, universal, and nutritious school to children in Swedish primary schools in the 1950s and 1960s. For this purpose, we use historical data on the rollout of the program across Swedish municipalities in the 1960s. We employ a difference-in-differences empirical design, where our identifying assumption is that exposure to school lunches is as good as random, conditional on birth cohort fixed effects and municipality fixed effects. We provide a number of specification checks that support this assumption.

Our results show substantial long-term economic benefits of the program. Those exposed to school lunches during their entire primary school period increased their life-time income by approximately 3 percent. This result is robust to several alternative specifications of the exposure variable and is similar in magnitude for both genders. The results also point to that earlier, and thereby longer, exposure leads to greater effects and that children from poor households gain most from the program, although children from all households except the richest benefit to some extent.

To shed light on the role of nutrition as a mechanism behind the income effect, we turn to data from the military enlistment and from the medical birth register, where we find that exposure to school lunches has large and positive effects on height and overall health. In addition, we obtain substantial effects on educational attainment that could explain half of the school lunch premium.

Our results emphasize the importance of childhood conditions and complement the growing literature that analyses the effect of interventions and circumstances at different phases of childhood. We contribute by analyzing the effect of a policy-driven change in nutrition in the phase between early childhood and adulthood where little evidence on effective policy interventions exist.

Although we study the part of the Swedish school lunch program that was rolled out during the 1950s and 1960s, we believe our results are still relevant for Western countries today. The program was introduced in a relatively wealthy country where school children did not face food insecurity or malnutrition. The school lunches rather changed the nutritional content of the meals and made pupils switch from less nutritious lunch bags or lunches at home. This is of relevance for many countries today that plan to improve, or have improved, the nutritional content of the food served in schools.

# References

- Aizer, A., Eli, S., Ferrie, J., & Lleras-Muney, A. (2016). The Long-Run Impact of Cash Transfers to Poor Families. *American Economic Review*, 106(4), 935–971.
- Allen, L. H. (1994). Nutritional influences on linear growth: a general review. European Journal of Clinical Nutrition, 48 Suppl 1, S75–89.
- Almond, D., Currie, J., & Duque, V. (2017). Childhood Circumstances and Adult Outcomes: Act II. Technical report, National Bureau of Economic Research.
- Almond, D., Doyle Jr, J. J., Kowalski, A. E., & Williams, H. (2010). Estimating marginal returns to medical care: evidence from at-risk newborns. *The quarterly journal of economics*, 125(2), 591.
- Almond, D., Edlund, L., & Palme, M. (2009). Chernobyl's subclinical legacy: prenatal exposure to radioactive fallout and school outcomes in sweden. *The Quarterly journal of economics*, 124(4), 1729–1772.
- Anderson, M. L., Gallagher, J., & Ritchie, E. R. (2017). School Lunch Quality and Academic Performance. Technical report, National Bureau of Economic Research.
- Barker, D. J. (1997). Maternal nutrition, fetal nutrition, and disease in later life. *Nutrition*, 13(9), 807–813.
- Belot, M. & James, J. (2011). Healthy school meals and educational outcomes. Journal of Health Economics, 30(3), 489–504.
- Bernstein, L. S., McLaughlin, J. E., Crepinsek, M. K., & Daft, L. M. (2004). Evaluation of the School Breakfast Program Pilot Project: Final Report. Technical Report N-04-SBP, U.S. Department of Agriculture, Food and Nutrition Service, Office of Analysis, Nutrition, and Evaluation, Alexandria, VA.
- Bhalotra, S., Karlsson, M., & Nilsson, T. (2017). Infant Health and Longevity: Evidence from A Historical Intervention in Sweden. *Journal of the European Economic Association*.
- Bharadwaj, P., Løken, K. V., & Neilson, C. (2013). Early Life Health Interventions and Academic Achievement. American Economic Review, 103(5), 1862–1891.
- Bhattacharya, J., Currie, J., & Haider, S. J. (2006). Breakfast of Champions?: The School Breakfast Program and the Nutrition of Children and Families. *Journal of Human Re*sources, XLI(3), 445–466.
- Black, S., Devereux, P., Lundborg, P., & Majlesi, K. (2015). Learning to Take Risks? The Effect of Education on Risk-Taking in Financial Markets. Technical Report w21043, National Bureau of Economic Research, Cambridge, MA.
- Brown, D. W., Kowalski, A. E., & Lurie, I. Z. (2015). Medicaid as an Investment in Children: What is the Long-Term Impact on Tax Receipts? Technical report, National Bureau of Economic Research.

- Bryan, J., Osendarp, S., Hughes, D., Calvaresi, E., Baghurst, K., & van Klinken, J.-W. (2004). Nutrients for Cognitive Development in School-aged Children. *Nutrition Reviews*, 62(8), 295–306.
- Burgess, S. (2016). Human Capital and Education: The State of the Art in the Economics of Education. IZA Discussion Papers 9885, Institute for the Study of Labor (IZA).
- Bütikofer, A., Mølland, E., & Salvanes, K. G. (2016). Introducing a Free Nutritious School Breakfast: Long-Term Impacts on Education and Adult Earnings. In Session 60: Children Berlin, Germany.
- Carlsson, M., Dahl, G. B., Öckert, B., & Rooth, D.-O. (2015). The Effect of Schooling on Cognitive Skills. *Review of Economics and Statistics*, 97(3), 533–547.
- Cattan, S., Kamhofer, D., Karlsson, M., & Nilsson, T. (2017). The Short and Long-Term Effects of Student Absence: Evidence from Sweden. Technical report, Lund University.
- Chay, K. Y., Guryan, J., & Mazumder, B. (2009). Birth cohort and the black-white achievement gap: The roles of access and health soon after birth. Technical report, National Bureau of Economic Research.
- Chetty, R., Hendren, N., & Katz, L. (2016). The Effects of Exposure to Better Neighborhoods on Children: New Evidence from the Moving to Opportunity Project. American Economic Review. Forthcoming.
- Currie, J. & Almond, D. (2011a). Chapter 15 human capital development before age five. volume 4 of *Handbook of Labor Economics* (pp. 1315 – 1486). Elsevier.
- Currie, J. & Almond, D. (2011b). Human capital development before age five. In *Handbook of Labor Economics*, volume 4 (pp. 1315–1486). Elsevier.
- Dahl, G. B. & Lochner, L. (2012). The Impact of Family Income on Child Achievement: Evidence from the Earned Income Tax Credit. American Economic Review, 102(5), 1927– 1956.
- Delange, F. (2000). The role of iodine in brain development. Proceedings of the Nutrition Society, 59(01), 75–79.
- Dunifon, R. & Kowaleski-Jones, L. (2004). Exploring the influence of the national school lunch program on children. Technical report, Institute for Research on Poverty.
- Evans, C. E. L. & Harper, C. E. (2009). A history and review of school meal standards in the UK. Journal of Human Nutrition and Dietetics, 22(2), 89–99.
- Feyrer, J., Politi, D., & Weil, D. N. (2017). The cognitive effects of micronutrient deficiency: Evidence from salt iodization in the united states. *Journal of the European Economic* Association, 15(2), 355–387.
- Figlio, D. N. & Winicki, J. (2005). Food for thought: the effects of school accountability plans on school nutrition. *Journal of Public Economics*, 89(2-3), 381–394.

- Fitzsimons, E. & Vera-Hernández, M. (2016). Breast feeding and the weekend effect: an observational study. BMJ open, 6(7), e010016.
- Författningssamling, S. (1946). Kungl. Maj:ts kungörelse angående statsbidrag till anordnandet av skolmåltider given Stockholms slott den 28 juni 1946. Nr 553.
- Fredriksson, P., Ockert, B., & Oosterbeek, H. (2013). Long-Term Effects of Class Size. The Quarterly Journal of Economics, 128(1), 249–285.
- Frisvold, D. E. (2015). Nutrition and cognitive achievement: An evaluation of the School Breakfast Program. Journal of Public Economics, 124, 91–104.
- Galer-Unti, R. et al. (1995). *Hunger and food assistance policy in the United States*. Garland Pub.
- Gluckman, P. D. & Hanson, M. (2005). The fetal matrix: evolution, development, and disease / Peter Gluckman, Mark Hanson. New York: Cambridge University Press.
- Gómez-Pinilla, F. (2008). Brain foods: the effects of nutrients on brain function. Nature Reviews Neuroscience, 9(7), 568–578.
- Grasgruber, P., Cacek, J., Kalina, T., & Sebera, M. (2014). The role of nutrition and genetics as key determinants of the positive height trend. *Economics & Human Biology*, 15, 81–100.
- Gundersen, C., Kreider, B., & Pepper, J. (2012). The impact of the National School Lunch Program on child health: A nonparametric bounds analysis. *Journal of Econometrics*, 166(1), 79–91.
- Hanes, S., Vermeersch, J., & Gale, S. (1984). The National Evaluation of School Nutrition Programs: program impact on dietary intake. *The American Journal of Clinical Nutrition*, 40(2 Suppl), 390–413.
- Heckman, J. J. (2007). The economics, technology, and neuroscience of human capability formation. *Proceedings of the national Academy of Sciences*, 104(33), 13250–13255.
- Hinrichs, P. (2010). The effects of the National School Lunch Program on education and health. Journal of Policy Analysis and Management, 29(3), 479–505.
- Hjort, J., Sølvsten, M., & Wüst, M. (2017). Universal investment in infants and longrun health: Evidence from denmark's 1937 home visiting program. Forthcoming in the American Economic Journal: Applied Economics.
- Hoddinott, J., Maluccio, J. A., Behrman, J. R., Flores, R., & Martorell, R. (2008). Effect of a nutrition intervention during early childhood on economic productivity in guatemalan adults. *The Lancet*, 371(9610), 411 – 416.
- Holmlund, H. (2008). A researcher's guide to the Swedish compulsory school reform. Discussion Paper 87, Centre for the Economic of Education, London School of Economics and Political Science, London.

- Hopkins, L. & Gunther, C. (2015). A Historical Review of Changes in Nutrition Standards of USDA Child Meal Programs Relative to Research Findings on the Nutritional Adequacy of Program Meals and the Diet and Nutritional Health of Participants: Implications for Future Research and the Summer Food Service Program. *Nutrients*, 7(12), 10145–10167.
- Hoynes, H., Page, M., & Stevens, A. H. (2011). Can targeted transfers improve birth outcomes? Journal of Public Economics, 95(7-8), 813–827.
- Hoynes, H., Schanzenbach, D. W., & Almond, D. (2016). Long-Run Impacts of Childhood Access to the Safety Net. American Economic Review, 106(4), 903–934.
- Hoynes, H. W. & Schanzenbach, D. W. (2015). US Food and Nutrition Programs. Working Paper 21057, National Bureau of Economic Research.
- Imberman, S. A. & Kugler, A. D. (2014). The Effect of Providing Breakfast in Class on Student Performance: The Effect of Providing Breakfast in Class. *Journal of Policy Analysis and Management*, 33(3), 669–699.
- Isen, A., Rossin-Slater, M., & Reed Walker, W. (2017). Every Breath You Take | Every Dollar You'll Make: The Long-Term Consequences of the Clean Air Act of 1970. *Journal* of Political Economy.
- Jacob, B. A., Kapustin, M., & Ludwig, J. (2015). The impact of housing assistance on child outcomes: Evidence from a randomized housing lottery \*. *The Quarterly Journal* of Economics, 130(1), 465–506.
- Jönsson, A. (1990). Skolk: en forskningsresumé. Skolöverstyrelsen.
- Jonsson, J. & Mood, C. (2013). Ekonomisk utsatthet och välfärd bland barn och deras familjer 1968–2010. Underlagsrapport 2013-5-51.
- Kremer, Michael Vermeersch, C. (2005). School Meals, Educational Achievement, and School Competition: Evidence from a Randomized Evaluation. The World Bank.
- Kristjansson, B., Petticrew, M., MacDonald, B., Krasevec, J., Janzen, L., Greenhalgh, T., Wells, G. A., MacGowan, J., Farmer, A. P., Shea, B., Mayhew, A., Tugwell, P., & Welch, V. (2007). School feeding for improving the physical and psychosocial health of disadvantaged students. In The Cochrane Collaboration & B. Kristjansson (Eds.), Cochrane Database of Systematic Reviews. Chichester, UK: John Wiley & Sons, Ltd.
- Lee, J. Y. & Solon, G. (2011). The fragility of estimated effects of unilateral divorce laws on divorce rates. *The BE Journal of Economic Analysis & Policy*, 11(1).
- Leos-Urbel, J., Schwartz, A. E., Weinstein, M., & Corcoran, S. (2013). Not just for poor kids: The impact of universal free school breakfast on meal participation and student outcomes. *Economics of Education Review*, 36, 88–107.
- Ludwig, J. & Miller, D. L. (2007). Does Head Start Improve Children's Life Chances? Evidence from a Regression Discontinuity Design. *The Quarterly Journal of Economics*, 122(1), 159–208.

- Lundborg, P., Nilsson, A., & Rooth, D.-O. (2014a). Adolescent health and adult labor market outcomes. Journal of health economics, 37, 25–40.
- Lundborg, P., Nilsson, A., & Rooth, D.-O. (2014b). Parental Education and Offspring Outcomes: Evidence from the Swedish Compulsory School Reform. American Economic Journal: Applied Economics, 6(1), 253–278.
- Maluccio, J. A., Hoddinott, J., Behrman, J. R., Martorell, R., Quisumbing, A. R., & Stein, A. D. (2009). The impact of improving nutrition during early childhood on education among guatemalan adults<sup>\*</sup>. *The Economic Journal*, 119(537), 734–763.
- McEwan, P. J. (2013). The impact of Chile's school feeding program on education outcomes. *Economics of Education Review*, 32, 122–139.
- Milligan, K. & Stabile, M. (2011). Do Child Tax Benefits Affect the Well-being of Children? Evidence from Canadian Child Benefit Expansions. American Economic Journal: Economic Policy, 3(3), 175–205.
- Millimet, D. L., Tchernis, R., & Husain, M. (2010). School nutrition programs and the incidence of childhood obesity. *Journal of Human Resources*, 45(3), 640–654.
- Mirtcheva, D. M. & Powell, L. M. (2013). National School Lunch Program Participation and Child Body Weight. *Eastern Economic Journal*, 39(3), 328–345.
- Neidell, M., Herzog, K., & Glied, S. (2010). The association between community water fluoridation and adult tooth loss. *American journal of public health*, 100(10), 1980–1985.
- Neovius, M., Teixeira-Pinto, A., & Rasmussen, F. (2008). Shift in the composition of obesity in young adult men in Sweden over a third of a century. *International Journal of Obesity*, 32(5), 832–836.
- Nilsson, J. P. (2009). The long-term effects of early childhood lead exposure: Evidence from the phase-out of leaded gasoline. *Institute for Labour Market Policy Evaluation (IFAU) Work. Pap.*
- Nilsson, J. P. (2017). Alcohol availability, prenatal conditions, and long-term economic outcomes. Journal of Political Economy, 125(4), 1149–1207.
- Oreopoulos, P. (2003). The long-run consequences of living in a poor neighborhood. *The quarterly journal of economics*, 118(4), 1533–1575.
- Paige, D. M. (1972). The School Feeding Program: An Underachiever. Journal of School Health, 42(7), 392–395.
- Pérez-López, F. R., Pasupuleti, V., Mezones-Holguin, E., Benites-Zapata, V. A., Thota, P., Deshpande, A., & Hernandez, A. V. (2015). Effect of vitamin d supplementation during pregnancy on maternal and neonatal outcomes: a systematic review and meta-analysis of randomized controlled trials. *Fertility and sterility*, 103(5), 1278–1288.
- Pollitt, E. & Gorman, K. S. (1994). Nutritional Deficiencies as Developmental Risk Factors. In C. A. Nelson (Ed.), *Threats to optimal development: integrating biological, psychological, and social risk factors*, number v. 27 in Minnesota symposia on child psychology. Hillsdale, N.J: L. Erlbaum Associates.
- Prentice, A. & Bates, C. J. (1994). Adequacy of dietary mineral supply for human bone growth and mineralisation. *European journal of clinical nutrition*, 48, S161–76; discussion S177.
- Schanzenbach, D. W. (2009). Do School Lunches Contribute to Childhood Obesity? The Journal of Human Resources, 44(3), 684–709.
- Schanzenbach, D. W. & Zaki, M. (2014). Expanding the School Breakfast Program: Impacts on children's consumption, nutrition and health. Technical report, National Bureau of Economic Research.
- Silventoinen, K. (2003). Determinants of Variation in Adult Body Height. Journal of Biosocial Science, 35(2), 263–285.
- SOU (1945). Betänkande om skolmåltider avgivet av 1941 års befolkningsutredning. Statens offentliga utredningar 1945:47, Socialdepartementet, Stockholm.
- SOU (1958). Statsbidrag till anordnandet av skolmåltider. Statens offentliga utredningar 1958:37, Ecklesiastikdepartementet, Stockholm.
- van den Berg, G. J., Lundborg, P., Nystedt, P., & Rooth, D.-O. (2014). Critical Periods During Childhood and Adolescence. Journal of the European Economic Association, 12(6), 1521–1557.
- Virgin, E. (1970). Skolmåltider förr och nu. Stockholm: Utbildningsförl.
- Werner, B. (2007). Growth in Sweden: surveillance of growth patterns and epidemiological monitoring of secular changes in height and weight among children and adolescents. PhD thesis, Stockholm. OCLC: 185218499.
- Wherry, L. R. & Meyer, B. D. (2016). Saving Teens: Using a Policy Discontinuity to Estimate the Effects of Medicaid Eligibility. *Journal of Human Resources*, 51(3), 556–588.
- WHO (2006). Food and nutrition policy for schools. Technical report, World Health Organization (WHO).
- Wolfers, J. (2006). Did Unilateral Divorce Laws Raise Divorce Rates? A Reconciliation and New Results. American Economic Review, 96(5), 1802–1820.
- Wyon, D. P. & Abrahamsson, L. (1997). An experimental study of the effects of energy intake at breakfast on the test performance of... International Journal of Food Sciences & Nutrition, 48(1), 5.



Figure 1: Cumulative number of municipalities introducing the school lunch program between 1959 and 1969.

Notes: The figure shows the cumulative number of municipalities that introduced the school lunch program between 1959 and 1969. See text for details.

Figure 2: Trends in income, schooling, and average municipality income for early, mid, and late adopting municipalities, and for municipalities that already had introduced the school lunch program by 1959.



Notes: Panels A-C show average income between ages 35 and 45 and years of schooling for individuals born between 1942 and 1965 in municipalities that introduced the school lunch program between 1959 and 1969 or who had introduced it already by 1959 (reform <1959). The different lines distinguish between individuals in early, mid, and late adopting municipalities of the school lunch program and between individuals in municipalities that already had introduced the program by 1959. Panel D shows average income at the municipality level between 1952 and 1966. Income is measured in 2010 prices.

Figure 3: Effect of years of exposure to school lunches on lifetime income



Notes: The figure plots estimates and 95 percent confidence intervals from a regression on the effect of years of exposure to school lunches on average lifetime income. The reference category is 0 years of exposure. Control variables include cohort fixed effects, municipality fixed effects, sex, whether born in Sweden, and whether the individual was exposed to 9 years of mandatory primary school. See text for details.



Figure 4: Effect of age at first school lunch exposure on lifetime income

Notes: The figure plots estimates and 95 percent confidence intervals from a regression on the effect of age at first school lunch exposure on average lifetime income. Control variables include cohort fixed effects, municipality fixed effects, sex, whether born in Sweden, and whether the individual was exposed to 9 years of mandatory primary school. See text for details.

Panel A: Height (males) Panel B: Height (females) 1.5 1.5 ŝ ŝ с Г ŝ Years of exposure Years of exposure Panel C: BMI (males) Panel D: BMI (females) N -1 4. Years of exposure Years of exposure Panel E: Overweight (males) Panel F: Overweight (females) 90. 80. .03 .03 -.03 -.03 -.06 -.06 Years of exposure Years of exposure Coefficient Lower 95% CI Upper 95% CI

Figure 5: Effects of years of exposure to school lunches on various health outcomes at enlistment

Notes: The figure plots estimates and 95 percent confidence intervals from regressions on the effect of years of exposure to school lunches on height, BMI, and overweight/obesity. Control variables include cohort fixed effects, municipality fixed effects, whether born in Sweden, and whether the individual was exposed to 9 years of mandatory primary school. See text for details.



Figure 6: Effects of years of exposure to school lunches on health, cognitive skills, and educational outcomes

Notes: The figure plots estimates and 95 percent confidence intervals from regressions on the effect of years of exposure to school lunches on health, cognitive skills, and educational outcomes. Control variables include cohort fixed effects, municipality fixed effects, sex (except for enlistment outcomes where only males were observed), whether born in Sweden, and whether the individual was exposed to 9 years of mandatory primary school. See text for details.





Notes: The figure plots estimates and 95 percent confidence intervals from an event study regression. In the graph, years of exposure coincide with corresponds to those that left school more than 1 year before the school lunches were introduced, etc. Negative event time thus measures the cohort distance in years from the first cohort that was treated with school lunches. In the graph, 1 year of school lunches corresponds to the oldest cohort exposed to school lunches, i.e., those who began the last grade when school lunches were introduced and who, thus, were only exposed for one year. cohorts. Zero years of exposure corresponds to belonging to the cohort that just had left school when school lunches were introduced. Minus 1 Positive years of exposure thus represent both the intensity of school lunch exposure and the distance in years from the first cohort that was treated with school lunches. Control variables include cohort fixed effects, municipality fixed effects, sex, whether born in Sweden, and whether the individual was exposed to 9 years of mandatory primary school. See text for details.

|               | Observations | Fraction |
|---------------|--------------|----------|
| Meals 0 year  | 124,987      | 0.082    |
| Meals 1 year  | 9,610        | 0.006    |
| Meals 2 years | 8,932        | 0.006    |
| Meals 3 years | 9,090        | 0.006    |
| Meals 4 years | 9,379        | 0.006    |
| Meals 5 years | 9,521        | 0.006    |
| Meals 6 years | 10,253       | 0.007    |
| Meals 7 years | 59,473       | 0.039    |
| Meals 8 years | 11,399       | 0.007    |
| Meals 9 years | 1,277,116    | 0.835    |
| Observations  | 1,529,760    |          |

Table 1: Distribution of school lunch exposure.

Notes: This table shows the distribution of school lunch exposure in our sample of individuals born between 1942 and 1965.

|                          | Mean     | SD    | Ν           |
|--------------------------|----------|-------|-------------|
| Outcomes                 |          |       |             |
| log(Earnings)            | 12.106   | 0.493 | 1,529,760   |
| Years of schooling       | 12.176   | 2.513 | 1,443,114   |
| University attendence    | 0.335    | 0.472 | 1,443,114   |
| Maternal labor supply    | 0.619    | 0.486 | 1,529,760   |
| Paternal labor supply    | 0.905    | 0.293 | 1,529,760   |
| Height (males)           | 178.958  | 6.511 | $625,\!630$ |
| Height (females)         | 166.671  | 5.853 | $463,\!157$ |
| BMI (males)              | 21.49    | 2.707 | $621,\!394$ |
| BMI (females)            | 22.762   | 3.419 | 424,573     |
| Overweight (males)       | 0.09     | 0.286 | $621,\!394$ |
| Overweight (females)     | 0.209    | 0.406 | $670,\!694$ |
| Perfect health           | 0.630    | 0.483 | $634,\!275$ |
| Cognitive skills         | 5.154    | 1.939 | $619,\!652$ |
| Smoking (females)        | 0.295    | 0.456 | 492,549     |
| log(Birth weight)        | 8.146    | 0.160 | $609,\!638$ |
| Low birth weight         | 0.082    | 0.275 | $609,\!638$ |
| Very low birth weight    | 0.015    | 0.120 | 609,638     |
| Background factors       |          |       |             |
| Birth year               | 1957.636 | 5.265 | 1,529,760   |
| Female                   | 0.487    | 0.5   | 1,529,760   |
| Swedish                  | 0.984    | 0.126 | 1,529,760   |
| Compulsory school reform | 0.875    | 0.331 | 1,529,760   |

| Table 2: | Descriptive | statistics. |
|----------|-------------|-------------|
|----------|-------------|-------------|

Notes: This table shows descriptive statistics for our sample of individuals born between 1942 and 1965. See text for details on the sample selection.

|                   | (1)              | (2)              | (3)              |
|-------------------|------------------|------------------|------------------|
|                   | Income           | Income           | Income           |
| Panel A           |                  |                  |                  |
| Years of exposure | 0.0035           | 0.0029           | 0.0027           |
| 1                 | $(0.0008)^{***}$ | $(0.0007)^{***}$ | $(0.0006)^{***}$ |
| Panel B           |                  |                  |                  |
| Meals 1 year      | 0.0001           | -0.0009          | -0.0012          |
|                   | (0.0052)         | (0.0052)         | (0.0052)         |
| Meals 2 years     | 0.0007           | -0.0005          | -0.0008          |
|                   | (0.0057)         | (0.0056)         | (0.0056)         |
| Meals 3 years     | 0.0003           | -0.0014          | -0.0017          |
|                   | (0.0055)         | (0.0054)         | (0.0054)         |
| Meals 4 years     | 0.0035           | 0.0015           | 0.0010           |
| ·                 | (0.0060)         | (0.0059)         | (0.0059)         |
| Meals 5 years     | 0.0101           | 0.0076           | 0.0070           |
|                   | (0.0064)         | (0.0063)         | (0.0063)         |
| Meals 6 years     | 0.0128           | 0.0098           | 0.0091           |
|                   | $(0.0066)^*$     | (0.0064)         | (0.0064)         |
| Meals 7 years     | 0.0132           | 0.0096           | 0.0086           |
| U U               | $(0.0053)^{**}$  | $(0.0051)^*$     | $(0.0050)^*$     |
| Meals 8 years     | 0.0205           | 0.0162           | 0.0153           |
| v                 | $(0.0068)^{***}$ | $(0.0065)^{**}$  | $(0.0065)^{**}$  |
| Meals 9 years     | 0.0296           | 0.0234           | 0.0218           |
| v                 | $(0.0077)^{***}$ | $(0.0064)^{***}$ | $(0.0063)^{***}$ |
| Observations      | 1,529,760        | 1,529,760        | 1,529,760        |
| Birth year FE     | YES              | YES              | YES              |
| Municipality FE   | YES              | YES              | YES              |
| Linear trends     | NO               | YES              | NO               |
| Quadratic trends  | NO               | NO               | YES              |

Table 3: Effect of school lunch exposure on ln(income).

Notes: This table shows DiD estimates of the effect of school meal exposure on (ln)lifetime income (average income between ages 25-65). Column 1 show results from our main DiD equation (1). Column 2 adds municipality-specific linear trends. Column 3 instead includes municipality-specific quadratic trends. In addition to birth cohort and municipality fixed effects, all regressions control for sex, whether born in Sweden, and whether the individual was exposed to 9 years of primary school. Panel A shows the effect of school meals measured as a continuous treatment between 0 and 9 years. Panel B shows the effect of school meals when measured through a set of dummy variables, with 0 years being the reference category. Robust standard errors in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

|                   | (1)<br>Income quartile 1 Income | (2)<br>Income quartile 2 | (3) (4)<br>Income quartile 3 Income quartile 4 | (4)<br>Income quartile 4 | ${}^{(5)}_{ m Men}$         | (6) Women |
|-------------------|---------------------------------|--------------------------|--|--------------------------|-----------------------------|-----------|
| Years of exposure | 0.0057                          | 0.0023                   | 0.0026   | 0.0015                   | 0.0031                      | 0.0040    |
| Observations      | $(0.0012)^{***}$ $380,015$      | $(0.0011)^{**}$ 380,017  | $(0.0011)^{**}$<br>380,016                     | (0.0018)<br>380,016      | $(0.0010)^{+++}$<br>780,163 | 739,901   |
| Birth year FE     | YES                             | YES                      | YES  | YES                      | YES                         | YES       |
| Aunicipality FE   | YES                             | YES                      | YES  | $\mathbf{YES}$           | $\mathbf{YES}$              | YES       |

Table 4: Heterogeneity analyses: effect of school lunch exposure on income by parental income quartile and gender.

Notes: This table shows DiD estimates of the effect of school lunch exposure on (log) lifetime income. Columns 1 to 4 shows the effect within quartiles of mean parental income (in 1968). Column 5 shows the effect for males and column 6 shows the effect for females. In addition to birth cohort and municipality fixed effects, all regressions control for sex (columns 1-4), whether born in Sweden, and whether the individual was exposed to 9 years of primary school. Robust standard errors in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

|  | (1)<br>I  | (2) Height   | (3)   | BMI (4)   | (5)<br>Overweig  | (6) (6) Overweight/obesity   |
|--|---|--|---|---|--|--|
|  | Males   | Females  | Males   | Females   | Males  | Females  |
| Panel A<br>Years of exposure   | 0.0678<br>(0.0141)***   | 0.0504<br>$(0.0158)^{***}$   | 0.0044<br>(0.0064)  | 0.0036<br>(0.0090)  | 0.0007 (0.0007)  | -0.0004<br>(0.0010)  |
| Panel B<br>Meals 1 year  | 0.1644  | 0.0469   | -0.0603   | 0.0387  | -0.0094  | -0.0009  |
| Meals 2 years  | (0.367)<br>0.0881   | (0.1632)<br>0.2588   | (0.453)<br>0.0159   | (0.1041)<br>0.0852  | (0.226)<br>0.0046  | (0.0125)<br>0.0015   |
| Meals 3 years  | (0.579)<br>$0.3126^{*}$   | (0.1836)<br>0.3489   | (0.838)<br>0.0758<br>(0.0758)   | (0.0994)<br>-0.0203<br>(0.10203)  | (0.555)<br>0.0049  | (0.0130)<br>0.0056   |
| Meals 4 years  | $(0.053) \\ 0.5861^{***} \\ (0.000)$  | $(0.1895)^*$<br>0.5489   | (0.258)<br>0.0668   | (0.1107) -0.0070  | (0.465)<br>0.0049<br>(0.704)   | (0.0129)   |
| Meals 5 years  | (0.000)<br>$(0.5577^{***})$   | (0.1817)<br>0.3874<br>(0.1795)**   | (0.341) -0.0112   | (0.1120)<br>0.0117  | (0.504)<br>-0.0021<br>(0.788)  | (0.0124)<br>-0.0031<br>(0.0130)  |
| Meals 6 years  | (0.000)<br>$(0.8974^{***})$   | 0.5511   | (0.036)   |   | 0.0028   | (0.0063)   |
| Meals 7 years  | (0.000)<br>$(0.6237^{***})$   | (0.1888) * * * * 0.4090  | (0.902)<br>(0.0484  | (0.1076)<br>-0.0061<br>(0.0876)   | (0.703)<br>0.0055<br>(0.333)   | (0.01035)<br>-0.0091<br>(0.010.0)  |
| Meals 8 years  | (0.000)<br>$(0.5328^{***}$  | (0.1430)<br>0.4121<br>(0.1757)**   | (0.0475<br>0.0475<br>0.0555   | (0.0010)<br>-0.0672<br>(0.1050)   | (266.0)  | (0.0104)<br>-0.0161  |
| Meals 9 years  | (0.003)<br>$0.7726^{***}$   | 0.6469   | (0.0355)  | (0.1030)<br>(0.0579)  | (0.909)<br>0.0050<br>(0.500)   | (0.110.0)<br>(0.0007<br>(0.110.0)  |
| Observations   | 625,630   | 463,157  | 621,394   | (0.1010)<br>424,573   | (0.300) $(21,394)$   | 424,573  |
| Birth year FE<br>Municipality FE   | YES<br>YES  | YES<br>YES   | YES<br>YES  | YES<br>YES  | YES<br>YES   | YES<br>YES   |
| Notes: This table shows DiD estimates of the effect of school 3-4 show effects on BMI, and columns 5 and 6 on overweight, in Sweden, and whether the individual was exposed to 9 yean a linear measure, whereas Panel B shows the effect of school errors in parentheses. *** $p \le 0.01$ , ** $p \le 0.05$ , * $p \le 0.1$ | D estimates of the effect of<br>d columns 5 and 6 on ov-<br>individual was exposed<br>Panel B shows the effect<br>\$<\$0.01, ** p\$<\$0.05, * | Notes: This table shows DiD estimates of the effect of school lunch exposure on various medium-term health outcomes. Columns 1 and 2 show effects on height (cm). Columns 3-4 show effects on BMI, and columns 5 and 6 on overweight/obesity (BMI>25). In addition to birth cohort and municipality fixed effects, all regressions control whether born in Sweden, and whether the individual was exposed to 9 years of primary school. Panel A shows the effect of one additional year of school meal exposure, that is, exposure as a linear measure, whereas Panel B shows the effect of school meals measured through a set of dummy variables, with 0 years being the reference category. Robust standard errors in parentheses. *** p\$<50.01, ** p\$<50.05, * p\$<50.1. | rious medium-term hes<br>n addition to birth coh<br>Panel A shows the effe<br>ough a set of dummy v | dth outcomes. Columns 1<br>ort and municipality fixed<br>ct of one additional year c<br>variables, with 0 years bei | and 2 show effects on he<br>effects, all regressions c<br>of school meal exposure,<br>ng the reference categor | sight (cm). Columns<br>ontrol whether born<br>that is, exposure as<br>y. Robust standard |

Table 5: The effect of school lunch exposure on height, BMI, and overweight/obesity.

|                                  |   |                                    | Full                            | Full sample                                    |  |                                 |
|----------------------------------|---|------------------------------------|---------------------------------|--|--|---------------------------------|
|                                  | $\begin{array}{c} (1) \\ \text{Perfect health} \\ (\text{males}) \end{array}$ | (2)<br>Cognitive<br>skills (males) | (3)<br>Smoking<br>(females)     | (4)<br>Schooling<br>(all)                      | (5) Maternal LMP                       | (6)<br>Paternal<br>LMP          |
| Panel A<br>Years of exposure     | 0.0041<br>(0.0013)***   | 0.0032 $(0.0026)$                  | -0.0013 $(0.0011)$              | 0.0294<br>$(0.0063)^{***}$                     | 0.0034<br>$(0.0008)^{***}$             | 0.0003<br>(0.0004)              |
| <i>Panel B</i><br>Meals 1 year   | 0.0043  | 0.0223                             | 0.0048                          |  | -0.0025                                | -0.0044                         |
| Meals 2 years                    | (0.0125)<br>0.0158  | (0.415)<br>$0.0455^{*}$            | (0.0131)<br>-0.0102             | $(0.0349)^{**}$<br>0.0803                      | (0.0061)<br>0.0021                     | (0.0039)<br>-0.0043             |
| Meals 3 years                    | (0.0121)<br>0.0124  | (0.066)<br>0.0359                  | (0.0136)<br>0.0095              | $(0.0428)^{*}$<br>0.1240                       | (0.0070) $(0.0021)$                    | (0.0040)<br>0.0023              |
| Meals 4 years                    | (0.0120)<br>0.0078  | (0.148)<br>$0.0465^{*}$            | (0.0142)<br>0.0090              | $(0.0392)^{***}$<br>0.1561                     | (0.0066)<br>0.0191                     | (0.0040)<br>-0.0043             |
| Meals 5 years                    | (0.0113)<br>0.0324<br>(0.0120)**  | (0.066)<br>0.0244                  | (0.0142)<br>-0.0073<br>(0.0140) | $(0.0457)^{***}$<br>0.1937<br>$(0.0400)^{***}$ | $(0.0016)^{**}$<br>-0.0016<br>(0.0070) | (0.0039)<br>-0.0043             |
| Meals 6 years                    | (0.0130)  | (0.348)<br>0.0371                  | (0.0142)<br>(0.0163)            | $(0.0490)^{***}$<br>0.2081                     | (0.0078)<br>0.0114                     | (0.0046)<br>-0.0009             |
| Meals 7 years                    | $(0.0127)^{**}$<br>0.0280<br>(0.0000)****                                     | (0.183)<br>$0.0535^{***}$          | (0.0137)<br>-0.0079             | $(0.0522)^{***}$<br>0.2786<br>$(0.0420)^{***}$ | (0.0080)<br>0.0187<br>(0.0000)***      | (0.0045)<br>-0.0007<br>(0.003c) |
| Meals 8 years                    | $(0.0093)^{++}$<br>0.0330<br>(0.0148)***                                      | (0.008)<br>(0.0235)                | (6010.0)<br>-0.0075             | (0.0429)                                       | (0.0074<br>0.0074<br>(0.0085)          | (0.0030)<br>-0.0010<br>(0.0049) |
| Meals 9 years                    | (0.0140)<br>0.0394<br>(0.01304)***  | (0.410)<br>(0.0390)                | -0.0026<br>-0.0026              | (0.0010) $(0.2770)$                            | (0.0009)<br>0.0280<br>//0.00093        | (0.0001)                        |
| Observations                     | (0.01384)   | (0.103)<br>619652                  | (0.0131) $492,549$              | $(0.0019)^{***}$<br>1,443,114                  | $(0.0083)^{+++}$<br>1,529,760          | (0.0041)<br>1,529,760           |
| Birth year FE<br>Municipality FE | YES<br>YES  | YES<br>YES                         | YES<br>YES                      | YES<br>YES                                     | YES<br>YES                             | YES<br>YES                      |

Table 6: The effect of school lunch exposure on health, cognitive skills, schooling, and labor supply

Notes: This table shows estimates of the effect of school lunch exposure on various potential mediators. Column 1 shows regressions on a dummy indicating perfect health, as assessed at military enlistment. Column 2 shows effects on cognitive skills, while column 3 shows effects on smoking among females. Column 4 shows effects on years of schooling, whereas columns 5 and 6 show effects on mothers' and fathers' labor supply, measured through having positive labor income in 1970. In addition to birth cohort and municipality fixed effects, all regressions control for sex (columns 4-6), whether born in Sweden, and whether the individual was exposed to 9 years of primary school. Panel A shows the effect of one additional year of exposure to school lunches. Panel B shows the effect of school meals measured through a set of dummy variables, with 0 years being the reference category. Robust standard errors in parentheses. \*\*\*  $p_s < 50.01$ , \*  $p_s < 50.01$ .

|                   | (1)              | (2)                   | (3)                   | (4)            | (5)       | (9)             |
|-------------------|------------------|-----------------------|-----------------------|----------------|-----------|-----------------|
|                   | Income           |                       | Income                | Height         | Height    | Perfect health  |
|                   |                  | First and second born | Second and third born | (males)        | (females) | (males)         |
| Years of exposure | 0.0040           | 0.0048                | 0.0040                | 0.0417         | 0.0333    | 0.0068          |
| ł                 | $(0.0009)^{***}$ | $(0.0015)^{***}$      | $(0.0023)^{*}$        | (0.0347)       | (0.0631)  | $(0.0035)^{**}$ |
| Observations      | 710,946          | 710,946               | 292,865               | 172,487        | 224,872   | 174,508         |
|                   |                  |                       |                       |                |           |                 |
| Family FE         | NO               | YES                   | YES                   | YES            | YES       | YES             |
| Birth year FE     | YES              | YES                   | YES                   | YES            | YES       | YES             |
| Municipality FE   | $\mathbf{YES}$   | YES                   | YES                   | $\mathbf{YES}$ | YES       | YES             |

| s models.       |
|-----------------|
| effects         |
| fixed           |
| Family f        |
| me).            |
| sure on ln(inco |
| re on           |
| expc            |
| lunch           |
| school          |
| of              |
| : Effect        |
| Table $7:$      |

of the number of years of school lunch exposure when restricting the sample to families where the first and second born children were born at most 5 years apart. Column 2 shows estimates from the same model as in column 1, this time with familiy fixed effects added. Column 3 shows estimates from re-running the model as in column 2, but restricting the data to second and third born children, born at most 5 years apart. Column 4 and 5 show estimates from family fixed effects regressions with height as an outcome, restricting the sample to first and second born siblings, born at most 5 years apart. Columns 6 show estimates from a family fixed regression with health at enlistment as an outcome, restricting the sample to first and second born siblings, born at most 5 years apart. In addition to birth cohort and municipality fixed effects, all regressions control for sex (columns 1-3), whether born in Sweden, and whether the individual was exposed to 9 years of primary school. Robust standard errors in parentheses. Standard Notes: The table provides robustness checks on our main DiD model of the effect of school lunch exposure on (log) lifetime income. Column 1 shows estimates of the effect errors clustered at the family level. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

|                   | (1)              | (2)              | (3)                   |
|-------------------|------------------|------------------|-----------------------|
|                   | (ln)Birth weight | Low birth weight | Very low birth weight |
| Years of exposure | 0.0000           | -0.0005          | -0.0004               |
| O1                | (0.0003)         | (0.0006)         | (0.0002)              |
| Observations      | 609,638          | 609,638          | 609,638               |
| Birth year FE     | YES              | YES              | YES                   |
| Municipality FE   | YES              | YES              | YES                   |

Table 8: Effect of school lunch exposure on the birth weight of children.

Notes: The table provides DiD estimates of the effect of school lunch exposure on the birth weight of children. Column 1 shows the effect of the number of years of school lunch exposure on log birth weight. Columns 2 and 3 show the effects on the probability of having at least one child born with low birth weight (<2500 grams) and very low birth weight (<1500 grams). In addition to birth cohort and municipality fixed effects, all regressions control for whether born in Sweden, and whether the individual was exposed to 9 years of primary school. Robust standard errors in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

|                   | (1)              | (2)              | (3)                       | (4)                          | (5)                               | (9)                                     |
|-------------------|------------------|------------------|---------------------------|------------------------------|-----------------------------------|---|
|                   | Labor income     | Income<br>35-45  | Income<br>School reform=1 | Income (males)<br>Enlistment | Income (females)<br>Height sample | Income (females)<br>Birth weight sample |
| Years of exposure | 0.0039           | 0.0023           | 0.005                     |                              |                                   | 0.0022                                  |
| 4                 | $(0.0010)^{***}$ | $(0.0009)^{***}$ | $(0.001)^{***}$           |                              |                                   | $(0.000)^{**}$                          |
| Observations      | 1,522,710        | 1,502,320        | 1,336,921                 | 720,408                      | 463,157                           | 609, 638                                |
| Birth year FE     | YES              | YES              | YES                       | $\mathbf{YES}$               | $\mathbf{Y}\mathbf{ES}$           | YÉS                                     |
| Municipality FE   | YES              | $\mathbf{YES}$   | $\mathbf{YES}$            | YES                          | YES                               | YES                                     |

| tests.     |  |
|------------|--|
| Robustness |  |
| 9: ]       |  |
| Table 3    |  |

Notes: This table shows the results of various robustness checks. Column (1) shows estimates from re-running our main specification, this time with labor earnings (excluding taxable benefits) as an outcome. Column (2) shows results using average earnings between ages 35-45 as an outcome. Column (3) shows results when only including individuals exposed to the 9-year schooling reform, see text for details. Column (4) shows estimates of the effect of school lunch exposure on earnings for male cohorts included in the military enlistment register. Column (5) shows the corresponding effect for females in the medical birth register (for those for which we observe birth weight). In addition to birth cohort and municipality fixed effects, all regressions control for sex (columns 1-3), whether born in Sweden, and whether the individual was exposed to 9 years of primary school. Robust standard errors in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

Tables and Figures For Online Publication

## Figure A.1: Example of a school lunch menu

Med utgångspunkt från de på sid. 27 föreskrivna typrätterna gives i det följande exempel på matsedel för sex veckor. Denna matsedel skall vara normgivande för måltidernas standard. Vid en skola kan man dock få ha en tre- eller fyraveckors matsedel.

## 1:a veckan.

| Måndag:             | Rökt korv, kokt blomkål eller annan grönsak, tomatsås,  |
|---------------------|---|
| Tisdag:             | potatis.<br>Blandad grönsakssoppa.<br>Smörgåspålägg: ost.   |
| Onsdag:<br>Torsdag: | Stekta fiskfiléer, citron, potatis. Frukt.<br>Ägg med spenatstuvning, tomat.                          |
| Fredag:<br>Lördag:  | Smörgåspålägg: leverpastej.<br>Leverbiff, råsallad, potatis.<br>Blandad risrätt, bönor. Frukt.        |
|                     | 2:a veckan.   |
| Måndag:             | Spenat- eller grönkålssoppa, ägg eller korv.  |
| Tisdag:             | Smörgåspålägg: ost.<br>Kalops, rödbetor, potatis. Frukt.<br>Grängslafst, ost, eller skirksås, potatis |
| Onsdag:<br>Torsdag: | Grönsaksfat, ost- eller skinksås, potatis.<br>Inkokt strömming, skarpsås, sallad, potatis.            |
| Fredag:             | Senapsbrynt fläskkorv, rotmos. Frukt.   |
| Lördag:             | Makaronipudding, smörsås, rårivna morötter.   |
|                     | 3:e veckan.   |
| Måndag:<br>Tisdag:  | Stuvade fiskbullar, ärter, potatis. Frukt eller morotsbit.<br>Skånsk kålsoppa.                        |
| i isuag.            | Smörgåspålägg: ost.   |
| Onsdag:             | Köttfärs, sås, tomat eller annan grönsak, potatis.  |
| Torsdag:            | Potatisgratin. Frukt.   |
| Freder              | Smörgåspålägg: korv.  |
| Fredag:<br>Lördag:  | Leverstuvning, äppel- och vitkålssallad, potatis.<br>Pannkaka. Frukt.                                 |
| Ũ                   | Source: SOU (1958). See text for details.   |

Source: SOU (1958). See text for details.

|  | (1)  | (2)  | (3)                   | (4)                   | (5)                   |
|--|--|--|-----------------------|-----------------------|-----------------------|
|  | Any diagnosis  | Type II diabetes   | Digestive conditions  | Visual impairment     | Congenital disorder   |
| Years of exposure                                | -0.0119 (0.0015)   | -0.0002 $(0.0001)*$  | -0.0059 (0.0006)***   | -0.0058<br>(0.000)*** | -0.0001 $(0.0003)$    |
| Observations<br>Birth year FE<br>Municipality FE | $\begin{array}{c} 635,773 \\ Y \mathrm{ES} \\ Y \mathrm{ES} \end{array}$ | $\begin{array}{c} 635,773 \\ \mathrm{YES} \\ \mathrm{YES} \end{array}$ | 635,773<br>YES<br>YES | 635,773<br>YES<br>YES | 635,773<br>YES<br>YES |
|  |  |  |                       |                       |                       |

| $\sim$          |
|-----------------|
| S.              |
| al              |
| male            |
| $\sim$          |
| pt              |
| men             |
| tn              |
| lis             |
|                 |
| ţ               |
| ъ.              |
| ons at en       |
| litions         |
| it              |
| H               |
| õ               |
| 2               |
| lth.            |
| ŝa.]            |
| Ъ               |
| nd              |
| ЧЦ              |
| re and l        |
| ur              |
| ò               |
| đ               |
| h exp           |
| д               |
| nc              |
|                 |
| 1               |
| 00              |
| ň               |
| š               |
| ••              |
| 1.1             |
| $\triangleleft$ |
| le              |
|                 |
| La              |

Notes: This table provides DiD estimates of the effect of school lunch exposure on various health conditions, diagnosed at military enlistment. In addition to birth cohort and municipality fixed effects, all regressions control whether born in Sweden and whether the individual was exposed to 9 years of primary school. Robust standard errors in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.01.

| Panel A<br>Years of exposure  | University attendence  | $^{(2)}$ Schooling   | (3)<br>University attendence   | (4)<br>Maternal LMP  |
|---|--|--|--|--|
| Donal R   | 0.0019 $(0.0009)**$  | $0.0294$ $(0.0062)^{***}$  | 0.0018<br>(0.0009)*  | 0.0026<br>$(0.0010)^{**}$  |
| Meals 1 year  | 0.0036   | 0.0760   | 0.0036   | -0.0012  |
| Meals 2 years   | (0.0010)   | (0.0349) $(0.0303)$  | (0.0053)<br>0.0011<br>0.0012   | (0.0079)<br>0.0025<br>20.0006)   |
| Meals 3 years   | (0.0001)<br>-0.0002<br>(0.0058)  | (0.0428)<br>(0.1239)<br>(0.0201)***  | -0.0002<br>-0.0002<br>-0.0058)   | (0.0090)<br>0.0035<br>(0.0008)   |
| Meals 4 years   | 0.0062   | (0.0391)<br>0.1560<br>0.0455)***   | 0.0062<br>0.0062   | (0.0090)<br>0.0225<br>(0.0109)**   |
| Meals 5 years   | (0.0072)   | (0.0455)   | (0.0072)<br>0.0085<br>0.06728  | (0.0103)   |
| Meals 6 years   | (0.0016)<br>0.0153<br>(0.0003)*  | (0.0461)   | (0.0076)<br>0.0153<br>(0.0089)*  | (0.0103)<br>(0.0027)   |
| Meals 7 years   | 0.0141<br>0.0141<br>0.0061)**  | (0.0010)<br>0.2786<br>(0.0490)***  | 0.0137<br>0.0137<br>0.0061)**  | (0.0154)<br>0.0154<br>(0.0086)*  |
| Meals 8 years   | 0.0153<br>0.0153<br>0.00050)*  | (0.0429)   | (0.00152<br>0.0152<br>0.0059)*   | 0.0105   |
| Meals 9 years   | (0.0090)*  | (0.0500)<br>0.2768<br>(0.0602)***  | (0.001)*(0.001)*   | (0.0109)<br>0.0221<br>(0.0106)**   |
| Observations<br>Birth year FE<br>Municipality FE<br>Linear trends   | 1,443,114<br>YES<br>YES<br>NO  | 1,443,114<br>YES<br>YES<br>YES   | 1,443,114<br>YES<br>YES<br>YES   | $1,513,596\\ YES\\ YES\\ YES$  |
| Notes: This table shows DiD of the effect of school lumexposure on the probability of entering college. Columns specific linear trends. Column 4 show the effects on margressions control for sex, whether born in Sweden, and a continuous treatment between 0 to 9 years. Panel B sh Robust standard errors in parentheses. *** $p<0.01$ , ** p | Notes: This table shows DiD of the effect of school lunch exposure on 'sxposure on the probability of entering college. Columns 2 and 3 show th specific linear trends. Column 4 show the effects on maternal labor sultegressions control for sex, whether born in Sweden, and whether the indi a continuous treatment between 0 to 9 years. Panel B shows the effect o a Robust standard errors in parentheses. *** $p<0.01$ , ** $p<0.05$ , * $p<0.1$ . | various educational outcomes a<br>the effects on years of schooling a<br>pply with municipality-specific<br>vidual was exposed to 9 years of<br>f school meals measured throug | Notes: This table shows DiD of the effect of school lunch exposure on various educational outcomes and maternal labor supply. Column 1 shows the effect of school lunch exposure on the probability of entering college. Columns 2 and 3 show the effects on years of schooling and college attendance in regressions where we control for municipality-specific linear trends. Column 4 show the effects on maternal labor supply with municipality-specific trends. In addition to birth cohort and municipality fixed effects, all regressions control for sex, whether born in Sweden, and whether the individual was exposed to 9 years of primary school. Panel A shows the effect of school meals measured as a continuous treatment between 0 to 9 years. Panel B shows the effect of school meals measured through a set of dummy variables, with 0 years being the reference category. Robust standard errors in parentheses. *** $p<0.01$ , ** $p<0.05$ , * $p<0.1$ . | ws the effect of school lunch<br>we control for municipality-<br>numicipality fixed effects, all<br>of school meals measured as<br>being the reference category. |

54

|                   | (1)                  | (2)                  | (3)                  | (4)                  | (5)                 | (9)                  | (2)                   |
|-------------------|----------------------|----------------------|----------------------|----------------------|---------------------|----------------------|-----------------------|
|                   | Mortality            | Morbidity            | Heart disease        | Mental disease       | Cancer              | Endocrine conditions | Digestive conditions  |
| Years of exposure | 0.0008               | -0.0001              | 0.0004               | 0.0000               | -0.0005             | -0.0000              | -0.0001               |
| Observations      | (0.000)<br>1,538,518 | (0.0003) $1,538,518$ | (1.0004) $1,538,518$ | (0.0003) $1,538,518$ | (0.003) $1,538,518$ | (0.0002) $1,538,518$ | (0.0004)<br>1,538,518 |
| Birth year FE     | YES                  | $\mathbf{YES}$       | YES                  | YES                  | YES                 | YES                  | YES                   |
| Municipality FE   | $\mathbf{YES}$       | $\mathbf{YES}$       | YES                  | YES                  | $\mathbf{YES}$      | YES                  | YES                   |

| mortality.               |
|--------------------------|
| and                      |
| morbidity and mor        |
| exposure on long-term mo |
| on                       |
| exposure                 |
| lunch                    |
| 0                        |
| schoe                    |
| $\operatorname{of}$      |
| Effect                   |
| A.3:                     |
| Table                    |

between 1995 and 2013). Column 1 shows the effect of the number of years of school lunch exposure on the probability of not surviving until 2013. Column 2 show setimates of between 1995 and 2013). Column 1 shows the effect of the number of years of school lunch exposure on the probability of not surviving until 2013. Column 2 show estimates of the effect of school lunch exposure on the probability of having any hospitalization. Columns 3-7 show regressions on specific health conditions as diagnosed in the hospitalization register between 1995-2013. In addition to birth cohort and municipality fixed effects, all regressions control for sex, whether born in Sweden, and whether the individual was exposed to 9 years of primary school. Robust standard errors in parenthese. \*\*\* p<0.01, \* p<0.05, \* p<0.1.